



Systematic Review **E-Cardiac Care: A Comprehensive Systematic Literature Review**

Umara Umar ^{1,*}, Sanam Nayab ¹, Rabia Irfan ¹, Muazzam A. Khan ² and Amna Umer ³

- ¹ School of Electrical Engineering and Computer Science (SEECS), National University of Sciences and Technology (NUST), Islamabad 44800, Pakistan
- ² Department of Computer Sciences, Quaid i Azam University, Islamabad 45320, Pakistan
- ³ Department of Computational Sciences, The University of Faisalabad (TUF), Faisalabad 38000, Pakistan
- Correspondence: uumar.msit19seecs@seecs.edu.pk

Abstract: The Internet of Things (IoT) is a complete ecosystem encompassing various communication technologies, sensors, hardware, and software. IoT cutting-edge technologies and Artificial Intelligence (AI) have enhanced the traditional healthcare system considerably. The conventional healthcare system faces many challenges, including avoidable long wait times, high costs, a conventional method of payment, unnecessary long travel to medical centers, and mandatory periodic doctor visits. A Smart healthcare system, Internet of Things (IoT), and AI are arguably the best-suited tailor-made solutions for all the flaws related to traditional healthcare systems. The primary goal of this study is to determine the impact of IoT, AI, various communication technologies, sensor networks, and disease detection/diagnosis in Cardiac healthcare through a systematic analysis of scholarly articles. Hence, a total of 104 fundamental studies are analyzed for the research questions purposefully defined for this systematic study. The review results show that deep learning emerges as a promising technology along with the combination of IoT in the domain of E-Cardiac care with enhanced accuracy and real-time clinical monitoring. This study also pins down the key benefits and significant challenges for E-Cardiology in the domains of IoT and AI. It further identifies the gaps and future research directions related to E-Cardiology, monitoring various Cardiac parameters, and diagnosis patterns.



1. Introduction

Following the available information as confirmed by the World Health Organization (WHO) [1], cardiovascular disease claims a large number of causalities across the globe [2] and is responsible for approximately 80% of sudden deaths. Moreover, in more than 15% of the deaths, cardiac arrhythmia is considered the chief reason. Thus, promoting cardiovascular health is vital and requires an overhaul of healthcare systems [3].

The rapidly expanding Internet of Things (IoT) [4] technology has the capability to monitor and control the critical human functions, irrespective of where the individual is located or what they are doing. Medical IoT (MIoT) is a cutting-edge technology that functions by exploiting the advantages of the Internet at a very affordable cost with minimum effort. The MIoT-based cardiac system guarantees monitoring the physical symptoms [5] of cardiac patients, such as temperature, Blood Pressure (BP), Oxygen Saturation (SPO2), Electrocardiogram (ECG), Heart Rate (HR) [6], and linked environmental parameters effectively and without any failure. The MIoT cardiac care framework is a customized paradigm that meets the requisite medical and safety standards of pervasive cardiac healthcare, including serious heart-related issues.

Various cardiac (heart) abnormalities can be detected through an Electrocardiogram (ECG) which is a medical testing platform that keeps track of electrical activity the heart generates as it contracts. An electrocardiograph is a device that records patient's ECG. An ECG is a valuable tool for identifying problems associated with heart rate or heart rhythm.



Citation: Umar, U.; Nayab, S.; Irfan, R.; Khan, M.A.; Umer, A. E-Cardiac Care: A Comprehensive Systematic Literature Review. *Sensors* **2022**, *22*, 8073. https://doi.org/10.3390/ s22208073

Academic Editors: Andrei Velichko, Dmitry Korzun and Alexander Meigal

Received: 30 August 2022 Accepted: 4 October 2022 Published: 21 October 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). It offers assistance to the physician in determining whether a patient is having a heart attack or has had one in the past. An ECG is usually the first option for a cardiac test because of its proven dependability. An ECG is helpful to determine if one's pulse is difficult to feel (bradycardia), or it is too fast (tachycardia) to count accurately. An ECG can also show heart rhythm irregularities, i.e., arrhythmia. The main types of arrhythmia are mentioned in Table 1.

Table 1. Various Types of Arrhythmias.

Types of Arrhythmia	Explanation
Tachyarrhythmias	A fast heart rhythm with a rate of more than 100 beats per minute.
Bradyarrhythmias	Slow heart rhythms that may be caused by disease in the heart's conduction system.
Supraventricular arrhythmias	Arrhythmias that begin in the atria (the heart's upper chambers). "Supra" means above; "ventricular" refers to the lower chambers of the heart, or ventricles.
Ventricular arrhythmias	Arrhythmias that begin in the ventricles (the heart's lower chambers).

Similarly, atrial fibrillation, atrial flutter, and premature or extra beats are the other types of cardiac issues. Figure 1 shows waveforms for different arrhythmia types. Atrial fibrillation refers to a rapid, disorganized, and irregular heart rhythm., while atrial flutter is an atrial arrhythmia generated by a fast circuit in the atrium. Compared to atrial fibrillation, atrial flutter is typically more organized and regular.



Figure 1. Types of arrhythmia.

A comprehensive review of E-Cardiology, which encompasses the Internet of Things, artificial intelligence, and cardiology could help understand the essential building blocks of an IoT-based cardiac care system and intelligent diagnosis of various cardiovascular diseases. It can also help to develop a complete picture of various hardware devices (sensors), AI techniques, and communication technologies adopted by the existing studies in the field of intelligent cardiac healthcare.

Following the introduction, Section 2 of this paper briefly discusses related works; Section 3 highlights the contributions made by this paper. Section 4 elaborates on the review methodology adopted for conducting the survey. Section 5 gives the outcomes of the selected studies with a detailed analysis of the research questions (RQs). This section is

further divided into four subsections. Section 6 contains a discussion. Section 7 summarises the conclusions.

2. Related Works

This section presents a brief explanation of the related surveys in the field of IoT-based cardiac healthcare.

The primary goal of the study [7] was to collect the latest facts, figures, and evidence on the use of preprocessing techniques for heart disease classification. The review study also summarised the impact of the most frequently used preprocessing tasks and techniques and the performance in the field of cardiology. This review paper covered the literature from 2000 to June 2019.

A survey on IoT and AI in healthcare was presented by [8] for 2007 to February 2018. The paper highlighted the top application classifications, which included wearables, sensor networks, connectivity options, and disease detection and treatment. This review identified gaps and provided future research directions related to technology and design. However, this survey analysed only three online databases.

A review article on data mining techniques frequently used in the field of cardiology until 2015 was presented in [9]. The performance comparison of various data mining models in cardiology were also discussed in this review paper.

The authors in [10] presented a survey on the Internet of Things (IoT) for healthcare using mobile computing. This systematic study investigated how mobile computing assisted IoT in a healthcare environment. Moreover, the intention of this paper was to analyse the impact of mobile computing on IoT technology in Smart hospitals and the field of healthcare. This study covered the literature between 2011 and 2019.

Another study [11] proposed a substantial review of various IoT applications in a life-saving environment, as well as various other fields in Smart cities. It also contrasted IoT with M2M and highlighted some drawbacks of IoT technology. This review article covered 2013 to 2018 through the Scopus database.

Another study [12] presented literature on (IoT) technologies and several projects for healthcare in 2018. This paper provided a review of primary medical IoT sensors and an overview of state-of-the-art IoT infrastructure essential for healthcare. It focused on the latest IoT technologies for healthcare services, such as cloud computing, big data, RFID, WSN, Bluetooth, Wi-Fi, and other vital medical sensors. However, this study lacks a systematic review.

The study [13] highlighted various IoT applications and was presented in 2022. The study focused on IoT adoption in Pakistan and France in 2020. This systematic study highlighted the barriers and possibilities for the implementation of IoT applications. It also indicated the influence of COVID-19 on IoT adoption in the healthcare domain.

The [14] systematic review discussed telemedicine and healthcare IoT (HIoT). It covered 146 articles between 2015 and 2020. The articles were divided into five categories after a technical analysis. In addition to the benefits and limitations of the selected methods, a comprehensive comparison of evaluation techniques, tools, and metrics was also included. This study presented a summary of healthcare applications of IoT (HIoT).

The discussion so far is limited to only a particular aspect of Smart healthcare/E-Cardiology and does not genuinely attempt to cover the domain holistically. When we say "entire domain", it means AI-based IoT, which encompasses preprocessing techniques and also various communication technologies. According to the deficiencies of the existing review papers, we provide a comprehensive systematic literature review for the following reasons:

- The latest research articles need to be covered to assess the current state of the art.
- The present studies do not cover all the aspects of E-Cardiology.

The following section highlights the contributions made by this review study, thus bringing novelty to this systematic review study.

3. Contributions

- 1. This review paper highlights the influence of IoT, communication technologies, AI models, and preprocessing techniques in cardiac healthcare using our review protocol. Moreover, this study covers the complete and latest infrastructure for E-Cardiology, including its benefits and challenges. Thus, this systematic review covers almost all aspects of E-Cardiology which have not been discussed before in such a comprehensive way under one umbrella.
- 2. The study presents the systematic analysis of the most recent studies (2016 to 2021) to investigate our formulated research questions.
- 3. This paper incorporates monitoring of vital CCU parameters, ECG analysis, and classification of various heart disorders, thus giving a thorough picture of E-Cardiology.
- 4. This review study provides recommendations and future guidelines for researchers and cardiologists as well.

The next section discusses the research methodology adopted for our SLR.

4. Review Methodology

A systematic literature review (SLR) paradigm is followed in this paper for reviewing papers from the most reliable resources, as shown in Figure 2. Principally, the research work, applications, and monitoring/detection techniques provided by AI-aided MIoT in cardiac care are considered. The primary studies have been then passed through a quality assessment process for the study analysis to produce the best fit results.



Figure 2. SLR protocol outline.

The following subsections briefly describe the detail of each step involved in our review protocol.

4.1. Defining Review Strategy

The application of medical IoT in cardiac care is a compelling field of study for the researchers, so the primary focus of this SLR was to formulate the research questions exploring how medical IoT is affecting cardiac care and the significance of artificial intelligence in the diagnosis and detection of various heart diseases.

The review questions in Table 2 indicate how MIoT and AI are contributing to cardiac healthcare systems in Smart hospitals.

No.	Review Question	Motivation
RQ 1	What are the vital hardware components/sensors used in E-Cardiac architecture for different CCU parameters?	The main focus of this question is to identify different types of sensors and their features most often used in IoT-based Cardiac Healthcare.
RQ 2	What are the most important communication technologies used in E-Cardiac Care?	The question aims to find the most commonly used communication technologies in MIoT-based Cardiology.
RQ 3	Which pre-processing techniques are used in E-Cardiology along with the most widely used AI classifiers/models?	This question is designed to explore the current work in the field of medical IoT accompanied by artificial intelligence in cardiac healthcare and to identify various classification and preprocessing techniques used for predicting cardiovascular diseases.
RQ 4	What are the significant issues and challenges in the current E-Cardiology?	This question investigates major benefits, and current challenges of IoT-based cardiac healthcare system.

Table 2. Review questions and their motivation.

4.2. Defining Search Strategy

Once the research questions were designed, the next step was to indicate and state the search strategy to be followed precisely. Therefore, the primary literature mentioned in Appendix A (Table A1) was identified using three search strings which were used in the five digital databases, namely IEEE Xplore, ACM Digital Library, SpringerLink, ScienceDirect, and Google Scholar. These are the most popular online data resources in the domain of computer science and information technology. Second, these digital libraries were used as sources for previous systematic literature reviews related to computer science and E-Cardiology [15].

Our search span included the period of 2016 to 2021. The criteria used for the selection of search terms or keywords is mentioned below [16]:

- The important terms were extracted from the research questions.
- Synonyms and alternate spellings were identified for the key terms.
- Keywords were identified from various books and relevant research articles.
- For synonyms or alternating spellings, the Boolean operator OR was used.
 - Boolean AND operator was used to interlink significant terms.

After the critical analysis of the key terms, three search strings were formed in order to extract the relevant information. These search strings were checked on each of the aforementioned databases by changing their patterns to retrieve the best relevant results. The three search strings are given in Table 3.

Table 3. Search strings used for data retrieval.

No. Search String

- 1 (internet of things OR IoT OR IoT-based OR smart health) AND (cardiac OR heart OR CCU) AND (monitoring OR detection OR diagnosis OR disease OR parameters)
- (intelligent OR artificial intelligence OR AI OR machine learning OR deep learning OR preprocessing OR reduction OR
 cleaning OR data mining) AND (cardiac OR heart OR ECG OR arrhythmia OR cardiovascular OR smart health OR healthcare
 OR smart healthcare OR cardiology) AND (technique OR methods OR classification OR algorithm)
- 3 (internet of things OR IoT OR IoT-based) AND (cardiac OR heart OR CCU OR smart health OR smart healthcare) AND (benefits OR advantages OR challenges OR issues OR disadvantages)

4.3. Inclusion and Exclusion Criteria

To identify and include studies relevant to answer the RQs, inclusion and exclusion criteria were developed as described in the section "Defining Review Strategy". To find the most appropriate publications, we defined the inclusion and exclusion criteria as mentioned in Table 4.

Table 4. Inclusion and exclusion criteria.

Inclusion Criteria	Exclusion Criteria
The papers published in English were chosen on priority.	The papers published in other languages were not selected.
The most recently published research papers, i.e., 2016 to 2021, were singled out for studies.	Gray literature was excluded from the study list.
Papers describing an overview of current approaches that implement modern tools and techniques in E-Cardiology were selected.	Papers not defining the topic appropriately were excluded.
The main aim was to target the primary studies such as original research papers.	Duplicated material was removed.

The authors evaluated each forthcoming paper to decide whether it should be included or excluded. The selection of papers was accomplished by following the three steps mentioned below.

The first step included the removal of duplicated and redundant papers; perusing the keywords, abstracts, and titles of research articles was the next step. Reading of full length research papers was carried out in the last step. Accordingly, the inclusion and exclusion criteria were implemented to their full effect. The articles that attracted difference of opinion were discussed and reviewed again by the authors, either using the full text or the partial text, until a consensus was achieved on an agreed-upon draft.

4.4. Quality Assessment Criteria

In this step, based upon the coherence and relevancy, we analyzed all the collected studies to address the defined research questions. A deep analysis of each paper was made, and based on our research questions, 134 papers were selected. Out of those 134 research papers, the papers having considerable citation count, appearing in good impact factor journals, and being delivered at highly ranked conferences were finally selected, thus leaving a total of 104 papers for the review, shown in Appendix A (Table A1).

4.5. Quantitative Analysis

The last step of our review protocol design was conducted to execute necessary statistical analysis on quantitative data. In this step, we quantitatively summarised and analyzed the results extracted from various sources such as conferences, journals, and book sections. Then, we carried out some quantitative statistical analysis of the findings to explore more about our research questions (RQs) and trends.

Figure 3 gives a thorough overview of our screening and assessment method for the statistical analysis of our literature. Five databases were chosen for the review, as illustrated in this figure. A total of 502 documents were chosen for review and analysis. The majority of papers were discovered to be duplicates. Thus, 203 records were eliminated before screening. Papers were removed for a few different reasons. Articles were chosen in the screening process based on a planned inclusion and exclusion strategy. Following the screening, 104 papers were chosen based on inclusion and exclusion standards. Identification

Screening



Included

Eligibilith

Figure 3. Flow diagram showing the screening process for the systematic review.

The next section highlights the "Outcomes" of this systematic review.

5. Outcomes

Google Scholars:35

Science Direct: 11 Springer:12 (n = 104)

ACM: 19 IEEE:27

5.1. RQ 1: What Are the Vital Hardware Components/Sensors Used in E-Cardiac Architecture for Different CCU Parameters?

The cardiac healthcare monitoring system in an IoT sphere encompasses the various IoT sensory modules and technologies attached to the patient, receiving sensory data, and sending data to the cloud for further monitoring, processing, and decision making. In an IoT-based cardiac healthcare monitoring system, the sensors, such as heart rate/pulse sensor, temperature sensor, blood pressure sensor, blood oxygen sensor, and ECG sensor, obtain sensory parametric values from the patient, transmit data through specific communication technologies to the cloud, apply machine learning practices to the learned parametric values, and generate alerts to the specialist suggesting timely action when warranted.

(i) Heart Rate Sensory Unit

Heart rate monitoring plays a crucial role in patient cardiac abnormalities diagnosis, detection, and classification. Several cardiac ailments and disorders occur due to a patient's high or low heart rate. Normal beats per minute (bpm) are 60–100. Less than 60 bpm is considered to be low and greater than 100 is considered to be high bpm. We discovered comprehensive studies that used several types of heart beat sensors for bpm monitoring. The studies [17,18] used heart beat pulse sensors to measure patient heart rates in a real-time environment. In this article [19], the KY 093 module was used to obtain heart rate values. Using a MAX30100 pulse oximeter, the authors in [20] collected heart beat information. The publications [21–23] utilized an ECG module AD8232 to obtain the patients heart rate data in real time for monitoring purposes. Table 5 mentions recent studies on heart rate sensors. Each heart rate sensor has its own set of properties. This table shows some important characteristics of several heart beat sensor variants such as pins, type, operating voltage, low current supply, accuracy, and so on.

(ii) Temperature Sensory Unit

Body temperature is an essential parameter for the development of cardiac healthcare monitoring. Various analog and digital temperature sensors are available for determining body temperature. Temperature can be measured in celsius or fahrenheit. Temperatures above 37.5 or 38.3 celsius are considered high. The temperature sensor LM35 is referenced in [24–27] for health monitoring. The authors in [18,28] used an 18DS20 sensor for temperature monitoring in a real-time environment. Table 6 shows several of the temperature sensors, along with descriptions. The LM35 and 18DS20 sensors are the most widely employed temperature sensors in the research studies that we analysed.

(iii) Blood Pressure (BP) Sensory Unit

BP monitoring is a fundamental biological measure for the detection and diagnosis of cardiac incongruities and anomalies. BP values can be obtained using various sensory units and devices. Systolic and diastolic values are captured by BP sensors or devices to be examined by a physician. Normal BP is less than 120/80 mmHg, while low BP, called "hypotension", is below 90/60 mmHg, and high BP, called "hypertension", is above 140/90 mmHg. Our research discovered a publication on E-Cardiology that dealt with cardiac patients' BP. In 2017 [29], a digital BP monitor (OMRONHBP1300) was used to monitor and automatically detect cardiac arrhythmia. The paper published in 2019 [30] examined predicting cardiac ailments in E-Cardiology using ECG, cholesterol, and BP. The MPX10 BP sensor was utilised in the 2020 publication [27] for patient health monitoring. Table 7 shows lists of sensors and devices utilized in the past few years for BP monitoring of cardiac patients, along with their comparable attributes.

The multiple modules of BP sensors and devices used in past studies, as well as different sensors of other cardiac parameters, are shown in Table 7.

Features Heart Rate/ Low **Pulse Rate** Operating Electrodes **Major Pins** INT Type BPM Supply Acc Sensors Voltage Configuration Current IR LED Ν 3.3 V to 5.0 V Υ N/A N/A Pulse Sensor [31] GND, Vcc, Signal N (Analog)

Table 5. Features of heart rate sensors/devices.

Llaget Data/	Features										
Fleart Kate/ Pulse Rate Sensors	Major Pins	INT	Туре	Operating Voltage	Operating BPM Voltage		Electrodes Configuration	Acc			
AD8232 [32]	GND, 3.3 V, Output, LO+, LO-, SDN	Y	IR LED (Analog)	3.6 V	Y	170 μA (typical)	Y	N/A			
KY-039 [33]	[33] GND, Vcc, Signal		IR LED (Analog)	5 V	Y	N/A	Ν	N/A			
Holter Device [34]	3/5/12 Electrodes	Y	Digital Device	N/A	Y	N/A	Y	N/A			
SpO2 Sensor device [35]	Fingertip Sensor	Y	IR LED (Analog)	N/A	Ŷ	N/A	N	±2% for SPO2, ±2 bpm for Pulse Rate			
MAX30100 Pulse Oximeter and Heart Sensor [36]	VIN, SCL, SDA, INTERRUPT, IRD, RD, GND	Y	Int IR LED, Photo Sensor	1.8 V and 3.3 V	Y	170 μA, (typical)	Ν	98.84% for SPO2, 97.11% for Heart Rate			

Table 5. Cont.

Acc—accuracy; BPM—beats per minute; GND—ground; INT—integrated IR; LED—infrared light-emitting diode; IRD—IR LED to driver; LO—leads off; N/A—not applicable; N—no; RD—red LED to driver; SCL—serial clock; SDA—serial data; SDN—shutdown control input; V—voltage; VCC—voltage common collector; VIN—voltage input; Y—yes.

(iv) Oxygen Sensory Unit

Blood oxygen can be monitored using several IoT-based blood oxygen sensory units, such as pulse oximetry sensors, to obtain oxygen saturation levels along with a patient's heart rate. Ready-made wearable devices are also available to measure blood oxygen saturation levels. Blood oxygen is measured in percentage. The normal blood oxygen saturation is 90 to 100%. The study [37] describes a pervasive healthcare monitoring service system that uses an SpO2 device to measure oxygen saturation. The MAX30100 pulse oximeter has proven to be useful in measuring blood oxygen levels in cardiac patients [20]. Our findings and the literature on IoT-based cardiovascular healthcare monitoring used the sensors mentioned in Table 8 to measure oxygen saturation. This table lists several important and common features of oxygen saturation sensors and devices, such as addressed parameters, voltage, type, accuracy, pins, range, and so on. Table 8 shows that blood oxygen sensors/devices are used for cardiac patients in very few studies.

(v) ECG Unit

ECG is the most crucial biological parameter for monitoring, detecting, predicting, and classifying cardiac irregularities and variations in the human heart. The ECG AD8232 module was used in the studies [21,22,38–40] to monitor ECG and detect cardiac anomalies in cardiovascular patients. Heart abnormalities were detected with the use of the ECG AD8233 module [41]. In Table 9, recent papers published on multiple ECG sensors and devices are mentioned along with their necessary and comparable features. Low supply current, electrodes, the sampling rate, right leg drive shut down, single supply operation, high pass filter, output, operating temperature, pins, and other features of various ECG modules are considered as the most notable attributes.

Table 10 shows detailed and comprehensive literature analysed to find IoT-based cardiovascular sensors and devices used in previous studies from 2016 to 2021. This table demonstrates that the majority of research on ECG has been conducted using the ECG AD8232 module to detect anomalies in cardiac patients.

Table 6. Features of temperature sensors/devices.

	Features											
Temperature Sensor	Туре	°C/°F or Both	Acc	Operating Voltage Range	Alarm Signaling	Major Pins	Measurement Range					
LM35 [42]	Analog	°C	0.5 °C Acc guaranteeable at +25 °C	4 V to 30 V	Ν	VCC, VOUT, GND	Range is -55° to +150 °C					
DS18B20 [43]	Digital	Both	±0.5 °C Acc from -10 °C to +85 °C	3.0 V to 5.5 V	Y	GND, DQ, VDD, NC	Range is 55 °C to +125 °C and 67 °F to +257 °F					
MCP9700 [44]	Analog	°C	±4 °C (max.), 0 °C to +70 °C	2.3 V to 5.5 V	Ν	Vout, Vcc, GND, NC	Range is $-40 \ ^{\circ}C$ to $+125 \ ^{\circ}C$					
TMP100 [45]	Digital	°C	± 1 °C (Typical) from -55 °C to 125 °C and ± 2 °C (Max) from -55 °C to 125 °C	2.7 V to 5.5 V	Y	ADD0, ADD1, ALERT, GND, SCL, SDA, V+	Range is –55 and +125 °C					

Acc—accuracy; ADD—address select; °C/°F—centigrade/fahrenheit; DQ—data in/out; GND—ground; N—no; NC—no connection; SDA—serial data; SDN—shutdown control; V—voltage; VCC—voltage common collector; VDD—power supply voltage; VOUT—output; Voltage Y—yes.

Table 7. Features of blood pressure	sensors/c	levices
-------------------------------------	-----------	---------

					Features					
BP Sensors	Freq	Range	Major Pins	Pressure Hysteresis	Lin	Supply Voltage	Full Scale Span	RT	Offset Stability	Acc
MPX10 Series Pressure Sensor [46]	N/A	0–10 kPa	GND, Vs, +Vout, –Vout	±0.1 typical	Min –1.0, Max 1.0	3.0–6.0 Vs	Min 20 mV, Max 50 mV	1.0 ms	±0.5% VFSS	N/A
Omron HBP- 1300 digital Device [47,48]	50/60 Hz	0 to 300 mmHg	Start/Stop, Mode, Last Reading (buttons)	N/A	N/A	100–240 V AC	N/A	N/A	N/A	Within ±3 mmHg
Typical BP Monitor Sensor [49]	N/A	0 to 258 mmHg	Tube, Pressure Cuff, Pressure Control Valve, Bulb	typical ±0.25%	typical ±0.25%	N/A	N/A	1.0 ms	N/A	±1 mmHg

Acc—accuracy; BP—blood pressure IR; kPa—kiloPascal's; LED—infrared light-emitting diode; mmHg—millimeters of mercury; ms—millisecond; N/A—not applicable; RT—response time; V—voltage; Vout—voltage output; Vs—power supply.

Oxygen Sensors (Oximeter)		Features										
	INT	Addressed Parameters	Power Supply Voltage	Туре	Acc SpO2	Acc PR	Major Pins	SpO2 Range	PR Range			
MAX30100 [36]	Y	HR, SpO2	1.8 V to 3.3 V	IR LED	99.62%	97.55%	VIN, SCL, SDA, interrupt, IRD, RD, GND	N/A	N/A			
SpO2 Sensor Device [50]	Y	HR, SpO2	D.C. 3.4 V ~D.C.4.3 V	IR LED	±2% (80–100%); ±3% (70–79%)	±2% bpm	N/A	35 to 100%	25 to 250 bpm			

Acc—accuracy; GND—ground; INT—integrated; IRD—IR led to driver; HR—heart rate; N/A—not applicable; PR—pulse rate; RD—red LED to driver; SDA—serial data; SDN—shutdown control; V—voltage; VIN—voltage input; Y—yes.

Table 9. Features of ECG sensor/devices.

		Features												
ECG Sensors/ Devices	INT	Single/ Multi Lead	Low Supply Current	Elec	SR	Right Leg Drive Shut Down	Single Supply OPER	HPF	Out	OPER TEMP	Major Pins			
AD8232 [32]	Ŷ	Single Lead	170 μA (typical)	2 or 3	360 HZ	N	2.0 V to 3.5 V	2 Poles	Rail to Rail	40 °C to +85 °C	GND, 3.3 V, OUT, LO-, LO+, ~SDN, RA, LA, RL			
Holter Device [34]	Y	Multi Lead	N/A	3, 5 or 12	125 HZ	N/A	one AAA battery	N/A	ECG Signal Rec on Monitor	+10 °C to +40 °C	Multiple Leads			
ADAS1000 [51]	Ŷ	Multi Lead	N/A	5 or 6	800 HZ	N/A	3.15 V to 5.5 V	N/A	Monitor	−40 °C to +85 °C	64 lead LQFP [52], 56 lead LFCSP (Both has diff. pins)			
AD8233 [53]	Y	Single Lead	50 A typical	2 or 3	N/A	Y	1.7 V to 3.5 V	2 Poles Adjustable HPF	Rail to Rail	−40 °C to +85 °C	20 pins (GND, VS+, REFIN, HP- SENSE, HP- DRIVE, SDN, AC/ DC, FR, etc.			
Shimmer 3 [54–57]	Y	Multi Lead	T60 μA Maximum	4	24 MHZ	N/A	450 mAh battery	N/A	On Windows PC and SQL	N/A	5 ECG pins, 5 EMG pins			

AC/DC—alternating current/direct current; EMG—electromyography; FR—fast restore; GND—ground; HPF—high pass filter; HPSENSE—high pass sense; HPDRIVE—high-pass driver; HZ—hertz; INT—integrated; LA—left arm; LO—leads off; MHZ—megahertz; NA—not applicable; N—no; OPER TEMP—operation temperature; PC—personal computer; RA—right arm; REFIN—reference buffer input; RL—right leg; SDN—shutdown control input; SQL—structure query language sampling rate; V—voltage; Vs—power supply terminal; Y—yes.

Table 8. Features of oxygen sensors/devices.

				Sensors Used			
Year	ECG Module	Temp Sensor	BP Sensor	Pulse/ HB Sensor	Oxygen Sensor	Other Sensor /Device	Integrated Sensor
2016 [17]	X	MCP9700	×	Pulse Sensor	×	x	×
2016 [58]	X	×	×	×	×	PCG Sensor	×
2016 [59]	×	x	x	×	x	×	Wrist band for HB & BP (DNNS)
2016 [19]	×	x	x	FingerTip- Optical Sensor for PPG	×	×	×
2016 [60]	×	×	×	×	×	Wearable Watch (PPG sensor)	×
2016 [61]	Galilio Board plateform for ECG (UB-MMNS)	×	×	×	×	×	×
2017 [37]	Holter Devices		(UB-MNNS)	Holter Device, SpO2 Device (UB-MNNS)	SpO2 Sensor Device (DNNS)	×	
2017 [62]	X	X	x	Pulse Sensor	x	×	×
2017 [19]	X	18DS20	X	KY-093	x	×	X
2017 [29]	X	×	OMRONH- -BP1300	PPG Sensor		External Defibillator	×
2017 [63]	(UB-MNNS)	(UB-MNNS)	(UB-MNNS)	HB Sensor	×	Alchol Sensor, EMG (MNNS)	x
2017 [64]	Wearable SOC ECG (MNNS)	×	×	×	×	×	×
2018 [65]	X	×	×	x	x	×	MAX30100 (SpO2, HB)
2018 [66]	X	×	×	Pulse Sensor	×	×	×
2018 [18]	X	×	×	Pulse Sensor	×	×	×
2018 [38]	ECG Module AD8232	X	X	Pulse Sensor	×	X	X
2018 [20]	ECG Module AD8232	×	×	MAX30100	MAX30100	x	×
2018 [24]	X	LM35	(UB-MNNS)	HB Sensor	×	X	X
2018 [67]	×	×	×	X	X	×	WWSN for ECG, BP, Respiratory
2019 [28]	X	18DS20	X	HB sensor	X	X	X
2019 [25]	Pulse Sensor	LM35	X	Pulse Sensor	X	X	X
2019 [26]	×	LM35	X	Pulse Sensor	×	X	X
2019 [40]	ECG Module AD8232	×	×	×	×	X	×

 Table 10. Sensors used in previous studies.

Table 10. Cont.

				Sensors Used			
Year	ECG Module	Temp Sensor	BP Sensor	Pulse/ HB Sensor	Oxygen Sensor	Other Sensor /Device	Integrated Sensor
2019 [21]	ECG Module AD8232	×	×	ECG Module AD8232	×	×	×
2019 [68]	X	×	×	×	×	Bio Sensors of hospital	×
2019 [69]	×	X	X	X	×	×	Watch for HB, CL, bp, (DNNS)
2019 [30]	ECG AD8232 Module	X	BP Cuff (UB-MNNS)	Heart Rate Monitor	X	Near Infrared Sensor for CL	×
2019 [70]	ECG AD8232 Module	×	×	×	×	×	×
2019 [71]	(UB-MNNS)	×	×	×	×	×	X
2020 [72]	(UB-MNNS)	×	×	×	×	×	X
2020 [73]	×	(UB-MNNS)	(UB-MNNS)	Pulse Sensor	×	×	X
2020 [74]	X	×	×	HB sensor	×	Alchohal Sensor (MNNS)	×
2020 [75]	×	X	X	X	x	×	MD, AC, ENV Sensors (MNNS)
2020 [27]	ECG Module AD8232	LM35	MPX10	Pulse Sensor	Pulse Sensor	x	x
2020 [76]	3 Lead VCG signals (MNNS)	X	X	×	X	X	×
2020 [22]	ECG Module AD8232	x	×	ECG Module AD8232	x	x	x
2020 [77]	ADAS1000	TMP100	X	×	X	×	X
2020 [78]	Multiple ECG devices (MNNS)	x	×	×	×	×	x
2020 [79]	Shimmer3 ECG Unit	×	×	×	×	×	×
2020 [23]	ECG Module AD8232	X	X	ECG Module AD8232	X	×	×
2021 [80]	×	X	(UB-MNNS)	(UB-MNNS)	X	Glucose Sensor (MNNS)	×
2021 [81]	Wearabale Smart ECG device (UB-DNNS)	×	×	×	×	×	×

2021 [41]

ECG Module

AD8283

Year	Sensors Used										
	ECG Module	Temp Sensor	BP Sensor	Pulse/ HB Sensor	Oxygen Sensor	Other Sensor /Device	Integrated Sensor				
2021 [83]	Ready made ECG Device (UB-DNNS)	X	X	×	x	AllCheck Device	x				
2021 [82]	Self Made Device for ECG (NNS)	×	x	×	×	x	×				
2021 [84]	Multiscale ECG from 3 Sensors (UB-MNNS)	×	×	Wearable HB Sensor (MNNS)	Respiratory Sensor (MNNS)	Optical Sensor (MNNS)	×				

Х

Table 10. Cont.

Х

Х

BP/bp—blood pressure; CL—cholesterol; DNNS—device name not specified; ECG—electrocardiogram; HB—heartbeat; HR—heart rate; INT—integrated; MNNS—module number not specified; PCG—phonocardiograph; PR—pulse rate; PPG—photoplethysmography; UB-DNNS—used but device name not specified UB-MNNS—used but module number not specified; **X**—The specified parameter is not addressed.

X

Х

5.2. RQ 2: What Are the Most Important Communication Technologies Used in E-Cardiac Care?

Communication technologies and protocols can be defined as a set of rules, technologies, semantics, equipment, and programs used to transfer, process, communicate, and receive information. Communication technologies and protocols vary depending upon the technology and network type devised, developed, or utilized. Some of the protocols and communication technologies are discussed in this section. The publications mentioned in Table 11 address the communication technologies and protocols used in previous selected studies for the development of E-Cardiology, monitoring, detection, and classification. BL is a wireless technology for short-range communication and exchanging data between mobile and fixed devices. BL has a transmission power of 1 mw-100 mw and a 1 Mbps data rate. Its data transmission range is 30 feet. The wearable healthcare monitoring devices (wearable fitness watches and pulse oximeters) may have the BL features integrated. BL technology was also employed in previous research [19,39,41,58,66,73] for data transmission for E-Cardiology. In prior literature on E-Cardiology monitoring, BL technology was determined to be the most commonly used technology. Ethernet is a wired communication networking protocol that can be used in local area networks (LANs), metropolitan area networks (MANs), and wide area networks (WANs). Ethernet allows communication through data cables. The publications [58,83] used an Ethernet-wired technology for connectivity support between various hardware modules implemented for cardiovascular disease diagnosis. One existing research study found that Ethernet-wired communication is rarely used in E-Cardiology. GSM is a cell-based or mobile communication modem that works as a mobile communication system. GSM technology is also used in E-Cardiology to send SMS messages or dial calls. GPS, which helps people to find their position on Earth, consists of networks of satellites and receivers or devices that determine location.

The communication technologies and protocols used in E-Cardiology in previous research studies and findings are detailed in Table 11.

Х

Year	Communication Technologies Used														
	BT	ETH	GSM	GPS	GPRS	MQTT	SMS	SMPP	ZB	WIFI	TCP/ IP	Sc. P	Cloud	SP/ PC	Internet
2016 [17]	Ν	Ν	Y	Y	Y	Ν	Y	Y	Ν	Y	Ν	Ν	Y	Y	Ν
2016 [58]	Y	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Y	Y	Y	Y	Ν
2016 [59]	Y	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Y	Ν	Y	Y	Ν	Ν
2016 [85]	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν
2016 [60]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν
2016 [61]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Y	Ν
2017 [37]	Y	Ν	Y	Ν	Y	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Y	Y
2017 [<mark>62</mark>]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν
2017 [<mark>19</mark>]	Y	Ν	Y	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν	Y	Y	Ν
2017 [29]	Ν	Ν	Y	Ν	Y	Ν	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2017 [<mark>63</mark>]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Y	Ν	Y	Y	Ν
2017 [<mark>64</mark>]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Y	Y	Y	Ν
2017 [39]	Y	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Y	Ν
2018 [65]	Ν	Ν	Y	Y	Y	Ν	Y	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2018 [66]	Y	Ν	Ν	Ν	Ν	N	Y	Ν	Ν	Y	Ν	Y	Y	Y	Ν
2018 [18]	Ν	Ν	Ν	Ν	Ν	N	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2018 [38]	Ν	Ν	Y	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν	Y	Y	Ν
2018 [20]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Y	Ν	Y	Y	Ν
2018 [24]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Y	Ν	Ν	Ν	Ν
2018 [67]	Y	Ν	Y	Y	Y	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Y	Ν
2019 [28]	Ν	Ν	Ν	Ν	Ν	N	Y	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν
2019 [25]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν
2019 [26]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Y	Y	Y	Ν	Ν
2019 [40]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Y	Y	Y	Ν
2019 [21]	Y	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν
2019 [68]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Y	Y	Y	Ν
2019 [69]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2019 [30]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Y	Y	Y	N
2019 [70]	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν
2019 [71]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Y	Ν
2020 [72]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν
2020 [86]	Ν	Ν	Ν	Ν	Ν	N	Ν	Ν	Ν	Y	Ν	Ν	Ν	Y	Ν
2020 [73]	Y	Ν	Y	Y	Y	Ν	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2020 [74]	Ν	Ν	Y	Ν	Y	N	Y	Ν	Y	Ν	Ν	Ν	Ν	Ν	Ν
2020 [75]	Y	Ν	Ν	Ν	Ν	N	Ν	Ν	Ν	Y	Y	Y	Y	Y	Ν
2020 [27]	Ν	Ν	Ν	Ν	Ν	N	Ν	N	Y	Y	Ν	Ν	N	Y	N
2020 [76]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	N

 Table 11. Communication technologies used in previous studies.

Year						Comn	nunicati	on Techn	ologie	s Used					
	BT	ETH	GSM	GPS	GPRS	MQTT	SMS	SMPP	ZB	WIFI	TCP/ IP	Sc. P	Cloud	SP/ PC	Internet
2020 [22]	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2020 [77]	Y	Ν	Y	Ν	Y	Ν	Y	Ν	Ν	Y	Y	Y	Y	Y	Ν
2020 [78]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2020 [79]	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν
2020 [23]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Y	Y	Y	Y	Ν
2021 [80]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Y	Y	Y	Ν
2021 [81]	Ν	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν
2021 [82]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Y	Y	Ν
2021 [83]	Ν	Y	Ν	Ν	Ν	Y	Ν	Ν	Ν	Y	Y	Y	Y	Y	Ν
2021 [84]	Ν	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Ν	Ν	Ν
2021 [41]	Y	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Ν	Y	Ν	Ν	Y	Y	Ν

Table 11. Cont.

BT—Bluetooth; GSM—global system for mobile communications; GPS—global positioning system; GPRS—general packet radio service; MQTT—message queuing telemetry transport; N—No; Sc.P—security protocols; SMS—short message service; SMPP—short message peer-to-peer; SP/PC—Smart phone/personal computer; TCP/IP—transmission control protocol/Internet protocol; WIFI—wireless fidelity; Y—yes.

5.3. RQ 3: Which Pre-Processing Techniques Are Used in E-Cardiology, along with the Most Widely Used AI Classifiers/Models?

RQ 3 is divided into two subsections. The first subsection investigates and compares various AI Models for the classification and prediction of CVD. This part explores various studies that use different machine learning and deep learning models for CVD prediction. Our study also provides a comprehensive explanation about the algorithms and methodologies used for prediction and classification techniques and the different datasets and performance metrics that we used to evaluate the models. Furthermore, the data preprocessing techniques used with different classifiers are also indicated in the second subsection below.

5.3.1. AI Classifiers/Models and E-Cardiology

The prediction of CVD is a much discussed topic of research in the realm of healthcare. AI-based prediction systems can be of great help in detecting disease at an earlier stage which can reduce risk associated with disease progression. The concept of AI is not new in cardiac electrophysiology with automated ECG interpretation. It has existed in some form or other since the 1970s [87].

Artificial Intelligence (AI) is the reflection of human cognitive functions from the surroundings acquired by applying algorithms, pattern matching, cognitive computing, and deep learning to achieve specific objectives [88]. The ongoing progress in AI, primarily in the sub-domains of machine learning (ML) and deep learning (DL), have caught the attention of physicians hoping to develop newly integrated, dependable, and potent methods for ensuring standard healthcare in the critical field of cardiology.

Machine learning (ML) is a subset of AI to "teach" computers to analyze huge datasets in a quick, accurate, and efficient manner by using complex computing and statistical algorithms [89]. Supervised ML is more successful in predicting survival compared to the traditional clinical risk scores [90].

The study [91] proved that the accuracy of disease prediction can be increased by using an unsupervised type of ML for obstructive coronary artery disease in nuclear cardiology.

Deep learning (DL) is a supervised ML methodology that relies on neural networks and is known for the automated algorithms required to extract meaningful patterns from data collections [92]. In the medical context, the most widespread deep learning algorithms are artificial neural networks (ANN), multilayer perceptron (MLP), convolution neural networks (CNN/ConvNet), recurrent neural networks (RNN), radial basis function network (RBFN), deep belief networks, and deep neural networks (DNN) [88]. Compared with traditional supervised ML, the real strength of DL is that it is an effective, powerful, and flexible approach to representing complicated raw input data that does not demand manual feature engineering. For instance, while addressing the issue of automated ECG interpretation, early conventional supervised ML techniques depended on human-defined ECG features. In contrast, the modern DL model extracts patterns within raw ECGs to detect sinus rhythm and various other arrhythmias with a performance that equals the result of any cardiologist [93].

The significant areas of cardiac healthcare that can benefit from ML/DL techniques are prognosis, diagnosis, classification, treatment, and clinical workflow. Table 12 presents an overview of different AI algorithms extracted from the literature review on heart disease diagnosis/classification.

 Table 12. Overview of common ai techniques used in E-Cardiology.

No.	AI Algorithm	Description	Strengths	Limitations
1	Principal Component Analysis (Unsupervised)	A method of dimensionality reduction which aims to compute principal components and makes data more compressible.	 Compute principal components Avoids data overfitting High variance, improved visualization Reduce Complexity 	 Low interpretability of principal components. Dimensionality reduction may result in information loss.
2	K-Means Clustering (Unsupervised)	Generates k number of centroids that help to define clusters of data.	 Ensures convergence Can warm-start the positions of centroids Easily adjusts to new examples Assists the doctors in making more accurate diagnosis 	 Not suitable for data varying in size and density Noise sensitive
3	Decision Tree (Supervised)	For classifying examples, a decision tree is an easy and simple representation.	 Easy to interpret Avoids over-fitting by pruning Less sensitive to outliers Requires less data cleaning 	1. Instability 2. Relatively inaccurate
4	K-Nearest Neighbor (Supervised)	Saves all available cases and allocates new cases based on a similarity measure.	 Easy to implement & understand Used for both classification & regression problems 	 Significantly slow as the data size increase Computationally expensive Requires high memory
5	Naïve Bayes (Supervised)	An easy probabilistic classifiers based on Bayes' theorem.	 Scalable Fast Used for real-time predictions Not requires large amounts of data 	 Assumes attributes are mutually independent Zero Frequency limitation
6	Random Forest (Supervised)	A set of decision trees, usually trained with the "bagging" technique. It performs classification as well as regression tasks.	 Used for prediction Resistant to noise and overfitting Flexible, can handle large datasets easily 	 Can take up lots of memory Not that interpretable
7	Support Vector Machine (Supervised)	Indicates hyperplane which separates classes, based on a similarity measure, can be used as a linear or nonlinear kernel.	 Fast Relatively memory efficient Works well with clear margin of separation between classes 	 Difficult to interpret Not suitable for large datasets May need normalization & scaling
8	Logistic Regression (Supervised)	The logistic paradigm can be used to model the probability of a certain class or event happening.	 Easy to implement and interpret Efficient to train 	 Performs poorly with large no. of variables Used to predict only discrete functions Not capture interactions auto- matically
9	Backpropagation (Supervised)	Backpropagation is a widely used algorithm for training feedforward neural networks. It is a reliable tool for increasing the accuracy of predictions.	1. Fast 2. Simple 3. Easy to analyze 4. Flexible	 Sensitive to noisy/complex data Performance of backpropagation depends on input data
10	Deep Learning (ANN) (Supervised)	Multilayered processing technique that mimics human neuronal structure. Different types of ANN are CNN or ConvNet , MLP, RBFN, RNN, etc.	1. No feature engineering 2. Learn complex functions 3. Enhanced Accuracy 4. Scalabale Model	 Requires extremely large datasets Intensive computational power Difficult to interpret Significant processing time

A comparative analysis of different AI techniques frequently used in Smart cardiology for the prognosis/diagnosis of various CVDs is given in Table 13.

As suggested by WHO, by 2030 almost 23.6 million individuals will die from heartrelated causes [94]. CVDs are the main cause, but they can be cured and prevented. To reduce the risk involved, analysis is fundamental. The difficult part is accurate diagnosis [95].

Table 14 summarises the most recent work performed in the field of artificial intelligence related to CVDs.

	AI Techniques Performance Metrics										
No.	Used in Smart Cardiology	Cl. Acc	Sensitivity	Specificity	Flexibility	Efficiency	Com. CPLX	Interpretability	Large Dataset Handling	Training Time	Noise Tolerance
1	DT	L	Y	Y	Y	Y	L	Н	Ν	S	Ν
2	RF	Н	-	-	Y	-	L	L	Y	F	Ν
3	NB	L	Y	-	Y	Y	L	Н	Y	F	Y
4	PCA-KNN	Н	Y	Y	-	-	-	L	-	F	Ν
5	SVM	Н	Y	Y	Y	Y	Н	L	Ν	S	Ν
6	LR	L	-	-	Y	Y	L	Н	Y	F	Ν
7	BP	Н	Y	Y	Y	-	-	Н	Y	-	Ν
8	DL (ANN)	Н	Y	Y	Y	Y	Н	L	Y	S	Y

Table 13. Comparative analysis of various AI techniques used for prognosis/diagnosis of CVDs.

PCA-KNN—principal component analysis with K-nearest neighbor; NB—naïve Bayes; RF—random forest; SVM—support vector machine; LR—logistic regression; DL—deep learning; ANN—artificial neural networks; BP—backpropagation; Y—yes; S—slow; F—fast; L—low; H—high; N—no.

Table 14. Summary of AI-methodologies and data preprocessing techniques identified for E-Cardiology, from different studies.

		Findings								
Ref #	Year	AI Methodology	Prognosis/ Diagnosis Task	Types of CVDs	Cardiac Parameter/s	Cardiac Dataset	Preprocessing Task	Data Preprocessing Techniques	Accuracy %	Complexity
[94]	2016	DT	Coronary Heart Disease	N/A	N/A	UCI	N/A	N/A	86.7	М
[96]	2016	BBNN	Arrhythmia	5	ECG	MIT-BIH	Feature Extraction	Hermit Basis Function	97	Н
[97]	2016	NN	Arrhythmia	5	ECG	MIT-BIH	Denoising Feature Extraction	DWT DWT + PCA DWT + ICA	98.91	Н
[98]	2016	PSO tuned SVM	Arrhythmia	12	ECG	MIT-BIH	Feature Extraction	DOST	99.18	Н
[99]	2016	NN	Arrhythmia	5	ECG	MITBIH	Feature Extraction	DOM	95	М
[100]	2016	RF	Arrhythmia	5	ECG	MIT-BIH	Feature Extraction	WPE	94.61	М
[61]	2016	SVM	Arrhythmia	2	ECG	CT	Feature Extraction	DWT	98.9	М
[101]	2016	Paired-CNN	Coronary Artery Calcification	N/A	ССТА	СТ	Feature Extraction	ConvNet	Sens. = 71	Н
[102]	2017	DL	Arrhythmia	N/A	ECG	MIT-BIH	Feature Extraction	AlexNet (DNN)	92	Н
[103]	2017	SVM	Arrhythmia	4	ECG	MIT-BIH	Denoising Feature Extraction	Multiresolution DWT	98.39	М
[104]	2017	RBF-NN	Arrhythmia	6	ECG	MIT-BIH	Denoising Feature Extraction	DWT EMD Features	99.89	Н
[102]	2017	DL	Arrhythmia	3	ECG	MIT-BIH	Feature Extraction	Transferred Deep Learning	92	Н
[105]	2018	SVM	Arrhythmia	3	ECG	CUDB VFDB	Feature Extraction	FFT	95.9	М
[106]	2018	SVM	Arrhythmia	5	ECG	MIT-BIH	Denoising Normalization Feature Extraction	Daubechies wavelets min-max Normalization PCANet	97.77	М
[107]	2018	Twin LS-SVM	Arrhythmia	16	ECG	MIT-BIH	Feature Extraction	Composite Dictionary (DOST + DST + DCT)	99.21	Н
[108]	2018	MPNN	Arrhythmia	3	ECG	MIT-BIH	Denoising Feature Extraction	Daubechies wavelet Multiresolution DWT MPNN	99.07	Н
[109]	2018	DL	Arrhythmia	5	ECG	MIT-BIH	Normalization Feature Extraction	Z-score normalization CNN and LSTM	98.10	Н
[110]	2018	DL	Arrhythmia	2	ECG	MIT-BIH	Feature Extraction	DNN	99.68	М
[111]	2018	DBN	Arrhythmia	5	ECG	MIT-BIH	N/A	N/A	95.57	Н
[91]	2018	DNN	Arrhythmia	N/A	CCTA	CT	Feature Extraction	DNN	Sens. = 82.3	Н
[71]	2019	CNN	Arrhythmia	4	ECG	CT MIT-BIH	Feature Extraction	CNN	94.96 95.73	М
[112]	2019	MPNN-BP	Heart Disease	5	ECG	UCI	N/A	N/A	97.5	Н
[93]	2019	DNN	Arrhythmia	12	ECG	СТ	Denoising Feature Extraction	DNN	ROC = 97 F1 = 83.7	Н
[68]	2019	Hybrid Model	Heart Disease	8	ECG, HR, BP	СТ	Data Cleaning Denoising Feature Selection	Numerical Cleaner Filter SFS	98	Н

						Findin	gs			
Ref #	Year	AI Methodology	Prognosis/ Diagnosis Task	Types of CVDs	Cardiac Parameter/s	Cardiac Dataset	Preprocessing Task	Data Preprocessing Techniques	Accuracy %	Complexity
[113]	2020	CNN-KCL	Myocarditis	N/A	ECG	ZAS	Outlier Anomaly K-means clustering	K-means clustering CNN	92.3	Н
[114]	2020	CNN	Myocardial Infarction	N/A	ECG	PTB	Data Augmentation Segmentation Feature Extraction	CNN	99.02	Н
[115]	2020	DNN	Arrhythmia	6	ECG	TNMG	N/A	N/A	F1 = 80 Spec. = 99	Н
[72]	2020	TWSVM	Arrhythmia	16	ECG	CT MIT-BIH	Feature Extraction	DWT	95.68	Н
[78]	2020	DHCAF MCHCNN	Arrhythmia	5	ECG, HR	CT MIT-BIH	Denoising Feature Extraction	Daubechies wavelet-4 HWT	91.4 93	Н
[116]	2021	E-D CNN-SVM	Myocardial Infarction	N/A	ECHO	HMC-QU	Featuring Engineering	CNN	80.24	Н
[117]	2021	RF, NB	Coronary Heart Disease	N/A	N/A	OR	N/A	N/A	83.85 (RF) 82.35 (NB)	М
[118]	2021	AI	Cardiac Amyloidosis	N/A	ECG	MC	Feature Extraction	DNN	90	N/A
[119]	2021	CNN	Heart Failure	N/A	N/A	СТ	Feature Selection	LASSO Regression	97	Н

Table 14. Cont.

H—high; M—medium; N/A—not available; DL—deep learning; MPNN-BP—multilayer perceptron neural network-backpropagation; RF—random forest; WPE—wavelet packet entropy; CNN—convolutional neural network; CL—K means clustering; ZAS—Z alizadeh; PTB—physikalisch technische bundesanstalt; NB—naive Bayes; FFT—fast fourier transform; DNN—deep neural network; SVM—support vector machine; NN—neural networks; BBNN—block-based neural network; MC—Mayo Clinic; HWT—haar wavelet transform; EMD—empirical mode decomposition; RBNN—radial basis function neural network; OCAD—obstructive coronary artery disease; DBN—deep belief networks; ROC—receiver operating characteristic curve; DHCAF—dynamic heartbeat classification; SFS—sequential forward transform; DOST—discrete orthogonal stockwell transform; CUDB—Creighton University database; VFDB—ventricular fibrillation database; LS—SVM-least square SVM; OR—online repository; DBN—deep belief networks; ROC—receiver operating characteristic curve; DWT—discrete wavelet transform; DHCAF—dynamic heartbeat classification and the probability of the pr

5.3.2. Data Preprocessing Techniques in E-Cardiology

This section identifies and evaluates studies that applied data preprocessing techniques in cardiac disease classification. Data Preprocessing (DP) in AI is a critical stage that enhances the quality of data to achieve meaningful insights and is the initial step in the development of an AI model. Conventionally, real-world data is not in an appropriate format and contains errors or outliers. It usually lacks specific attribute values/trends, thus resulting in an inadequate AI model. Data preprocessing solves this problem by cleaning and organizing raw data to tailor it to the needs of building and training AI models. Hence, data preprocessing in AI is a data mining approach that reshapes raw data into a readable format that is readily available for an AI model to meet the high standards of performance [120]. Consequently, the algorithm can easily interpret the data's features.There are four primary ways of data preprocessing, i.e., (1) data cleaning, (2) data integration and formatting, (3) data transformation, and (4) data reduction.

Different preporcessing techniques used in past studies for diagnosing heart disease and other types of arrhythmia are also mentioned in Table 14, along with AI models. This table also lists the task performed by the preprocessing technique.

5.4. RQ 4: What Are the Major Issues and Challenges in Current E-Cardiology?

After conducting comprehensive research, we identified some significant benefits and major challenges in the field of MIoT to answer our RQ 4. These challenges and benefits have been emphasized on the basis of studies conducted by different researchers in the domain of MIoT and E-Cardiology. Based upon selection and rejection criteria, only valid and reliable papers were selected, as mentioned earlier. We incorporated only the latest benefits and challenges that were found to be unique in the domain of IoT and AI regarding



healthcare and cardiology. A pictorial representation of these benefits and challenges is shown in Figure 4.

Figure 4. Benefits and challenges of E-Cardiology.

5.4.1. Benefits of E-Cardiology

Internet of Things (IoT) develops a linkage between "things", such as devices, gadgets, vehicles, and sensors. Likewise, the medical Internet of Things (MIoT)-based cardiovascular healthcare system monitors the physical symptoms [5] of cardiac patients at a very reasonable cost. These physical symptoms include temperature, blood pressure (BP), SPO2, and heartbeat, along with ECG [6] and associated numerical measurements. Significant benefits of IoT-based cardiology from various studies are noted in Table 15.

....

	_		Findings	
Ref #	Year	Key Challenges and Barriers of E-Cardiology with IoT	Data Related Issues	Benefits of IoT-Based E-Cardiology
[121]	2016	Security, Interoperability Unintended Behavior, Device Vulnerability	Privacy Consistency Integration	Cost Reduction Clinical Continuity Quality Life, Telemedicine
[122]	2016	Security, Interoperability Complexity, Scalability Device Vulnerability	Privacy	Cost Reduction Clinical Continuity Automation, Time Saving
[123]	2017	Security Energy Consumption Network Latency Intelligence in Medical Care System Predictability	Privacy Real-Time Processing	Cost Reduction Clinical Continuity, Automation, Time Saving, Quality Life, Telemedicine
[124]	2018	Security, Interoperability Energy Consumption Network Latency	Privacy	Ubiquitous Access, Quality Life Cost Reduction, Time Saving Reduced Hospital Visits
[11]	2019	Security, Interoperability Energy Consumption Internet Bandwidth	Privacy	N/A

Table 15. Key issues and major benefits of IoT-based cardiology.

Table 15. Cont.

		Findings								
Ref #	Year	Key Challenges and Barriers of E-Cardiology with IoT	Data Related Issues	Benefits of IoT-Based E-Cardiology						
[125]	2019	Security Heterogeneity	Privacy, Reliability, Utility Validity, Generalizability Integrity, Objectivity, Data Overload, Completeness, Relevance	Personalized, Predictive, Participatory, Preventative, Persuasive, Perpetual, Programmable (7P)						
[126]	2019	Security Unintended Behavior	Privacy, Confidentiality	Ubiquitous Access Cost Reduction Clinical Continuity Improved Accuracy Quality Life, Telemedicine						
[127]	2019	Security Scalability	Privacy Data Overload	Ubiquitous Access Cost Reduction, Clinical Continuity Automation, Time Saving Quality Life Telemedicine						
[128]	2019	Security Context-aware Computing Interoperability Energy Consumption	Privacy	N/A						
[129]	2020	Unobtrusiveness Energy Consumption Quality of Service Scalabilty, Fixation Patient Indentification Body Impact on Signal Propagation	Reliability Integrity Data Protection Data Representation Accuracy	N/A						
[130]	2020	Energy Consumption, Storage Patient's discomfort caused by Sensors	Privacy Data Overload Noise	Flexibility Clinical Continuity Remote Monitoring						
[131]	2020	Security	Privacy Confidentiality Integrity, Data Loss Aval1abilty Compromise	Remote Monitoring Cost Reduction, Time Saving Better Diagnostics Improved Clinical Infrastructure						
[132]	2020	Security Mobility Heterogeneity Legal Aspects	Privacy	N/A						
[133]	2021	Security, Scalability	N/A	Efficient, Cost Effective						
[134]	2021	Security Scalability Interoperability, Energy Consumption, Low Latency Tolerance	Privacy Computational Intensity	Ubiquitous Access Time Saving, Cost Reduction Telemedicine, Quality Life Clinical Continuity Easy Usge						

A comparison of some key factors in Smart cardiac care are shown in Table 16. *Artificial intelligence (AI)* is another significant aspect of E-Cardiology. Until now, AI in cardiac electrophysiology has exhibited promising results. Primary advantages of AI-based cardiology are discussed in Table 17.

 Table 16. Comparison of the existing evaluation factors in the Smart cardiac care.

Ref #	Year	Security	Privacy	Complexity	Integration	Reliability	System Predictability	Interoperability	Scalability	Heterogeneity	Energy Consumption	Network Latency
[122]	2016	1	1	1	1	×	×	1	×	x	×	X
[121]	2016	1	✓	×	1	×	×	1	×	×	×	×
[135]	2017	1	×	×	×	×	1	X	×	×	1	×
[123]	2017	1	1	×	×	×	✓	×	×	X	1	1
[124]	2018	1	✓	×	×	×	×	1	×	×	1	1
[11]	2019	1	1	×	×	×	×	✓	×	×	1	×
[126]	2019	1	1	×	×	×	×	×	×	×	×	×
[127]	2019	1	1	×	×	×	×	×	1	×	×	×
[125]	2019	1	1	×	1	1	×	×	×	1	×	×

Ref #	Year	Security	Privacy	Complexity	Integration	Reliability	System Predictability	Interoperability	Scalability	Heterogeneity	Energy Consumption	Network Latency
[128]	2019	1	1	×	×	×	×	1	×	×	1	X
[129]	2020	×	1	×	1	1	×	×	1	×	1	×
[130]	2020	×	1	×	×	×	×	×	×	×	1	×
[131]	2020	1	1	×	1	×	×	×	×	×	×	×
[132]	2020	1	1	×	×	×	×	×	×	✓	×	×
[133]	2021	×	1	×	×	×	×	×	1	×	×	×
[134]	2021	1	1	1	×	×	×	1	1	×	1	1

Table 16. Cont.

✓—parameter mentioned; **メ**—parameter not mentioned.

Ref #	N	Findings								
Kei #	iear	Key Challenges and Barriers for E-Cardiology with AI	Benefits of AI in E-Cardiology							
[136]	2016	Safety and transparency Algorithmic fairness and biases, Complexity Data privacy and information security Need for infrastructure, High quality data Public perceptions about AI, Informed consent	Better diagnosis, better services Improves quality of services Time saving Reduced treatment cost							
[137]	2019	Fitting confounders accidentally versus actual signal, Generalizability, Algorithmic bias Possibility of adversarial attack Logistical challenges in deploying AI systems Robust and rigorous quality assurance Traditional reluctance to switch from existing model to AI model in healthcare Algorithmic accountability To develop a relation between physicians and human-centered AI tools	N/A							
[138]	2019	Data privacy, Accountability Algorithmic bias Adaptability, Complexity	Improved healthcare Better diagnosis High accuracy							
[139]	2019	Privacy and discrimination Dynamic information and consent Transparency and ownership	Speedy imaging, Increased efficiency Greater insight into predictive screening Decreased healthcare cost							
[140]	2020	Respect for autonomy Beneficence Non-maleficence and justice	Lower cost Improved diagnosis and treatment							
[141]	2021	Safety, Privacy and security threats Ethical challenges Regulatory and policy challenges Availability of quality data and Lack of data standardization Distribution shifts Upgrading hospital infrastructure	Disease prediction and diagnosis Better image interpretation Real-time monitoring							
[142]	2021	Sometimes data reflects inherent biases and disparities Huge dataset requirement Patient's confidentiality Potential to be detrimental	Improved decision making Improved precision and predictability Intraoperative guidance via video							

Table 17. Key challenges and primary benefits of AI-based cardiology.

5.4.2. Challenges of E-Cardiology

Internet of Things (IoT) comes with various challenges as shown in Table 15. In addition to these challenges, ref. [129] proposed some other vital challenges, such as fixation of sensors, body impact on signal propagation, and synchronization, that may affect critical health services such as cardiac care. The MIoT-based healthcare systems are expected to produce a vast amount of data. Moreover, these sensors and devices are linked through networks, thus enabling real-time transmission of data. Therefore, hackers may attempt to target it. Moreover, the timely availability of medical data will affect the patient's life. Consequently, it is crucial to have real-time information with lower latency over the network [124].

Artificial intelligence (AI) has brought a revolution in the field of healthcare. It has especially made a great contribution in the domain of cardiac care, such as timely prediction and diagnosis of cardiovascular diseases (CVDs), ECG analysis, and arrhythmia

classification. However, despite all these milestones, it carries some challenges. Table 17 describes critical issues and challenges of AI-based cardiology from different past studies.

It is hoped that these challenges can be met and that through MIoT and AI we can achieve new levels of technical and medical standards in the field of cardiac healthcare.

6. Discussion

Figure 5 shows the distribution of the selected studies (104 papers) according to our four research questions (RQs). The pie chart shows that 47 studies overlap RQ1 and RQ2, whereas 33 studies address RQ3 and 24 address RQ4.



Figure 5. Statistical analysis of reviewed papers in terms of formulated RQs.

The outcomes of this systematic literature review suggest a noticeable increase in the research conducted in IoT using artificial intelligence in E-Cardiology. The research activities also incorporate monitoring CCU parameters, ECG analysis, and diagnosis/classification of various heart diseases. This extensive review study reveals that IoT sensors utilized for E-Cardiology are based upon analog sensors, digital sensors, and different wearable sensors or device modules. For fitness tracking and health monitoring activities, a variety of ready-made and wearable watches are also available. E-Cardiology patients can use a variety of wearable gadgets to track various cardiac healthcare characteristics. To establish expert E-Cardiology systems, several communication technologies and protocols for transmissions over a defined range need to be instituted and maintained. Our findings on communication technologies and protocols for the implementation of IoT-based cardiovascular healthcare monitoring and detection systems show that the following technologies provide viable tools to use in MIoT systems: Bluetooth (BL/BLE), ethernet, global system for mobile (GSM), global positioning system (GPS), global packet radio service (GPRS), message queuing telemetry transport (MQTT), short message service (SMS), email, zigbee, transmission control protocol/Internet protocol (TCP/IP), wireless fidelity (Wifi), security protocols, broadband/Internet, cloud, Smart cars, Smart phones/computers, etc. Our findings show that deep learning is often used in cardiac imaging procedures, particularly in echocardiography [88]. Furthermore, CNNs have been evaluated and found useful in calculating coronary artery calcium in cardiac CT angiography [101]. Though deep learning techniques are garnering attention in the field of Smart cardiology, we may infer that instead of depending on a particular AI model, hybrid techniques are expected to

produce better results. From the results of our study, it may also be inferred that SVM was more frequently used in cardiac care than was deep learning. However, in recent years, deep learning has emerged as a more powerful and reliable tool for the detection and diagnosis of various heart diseases. It was also noted that data reduction appears to be a major concern of researchers when applying data mining/data preprocessing approaches to predict CVD. This literature review includes 24 studies devoted to major issues and challenges in E-Cardiology. These studies conclude that a major benefit of MIoT in cardiac healthcare is that it generates timely and accurate data, which results in better healthcare outcomes. One vital advantage of using ML techniques is the ability to fuse various types of data [143]. We also assessed the results and noted that mIoT devices do not possess the requisite data protocols and standards [131]. Therefore, many issues must be addressed to ensure IoT privacy and security [144,145], which is one of the major challenges of the IoT era. Moreover, the continuous monitoring of critical indicators requires reduced energy consumption and a longer battery life [146] to prevent a break of communication. This is also one of the significant challenges of MIoT. The medical data gathered using MIoT is seldom standardized and often fragmented. The data in legacy IT systems are usually generated with incompatible formats. Thus, the great challenge of interoperability needs to be addressed as well [124,147]. However, to move forward in the field of MIoT, fearing AI is no option.

Instead, we should work toward the smooth digitization of healthcare infrastructure [148]. Obviously, various benefits of AI cannot be implemented and utilized correctly without integrating AI into clinical decision making effectively and responsibly [149].

Figure 6 refers to the research work conducted based on the number of papers per annum. It shows the year-wise trends in publications in the field of E-Cardiology. It also indicates that the maximum number of papers selected for this survey was from 2019.

6.1. Gaps, Future Recommendations

In a future work, all the vital cardiac parameters can be combined with an intelligent cardiac care unit (CCU) to develop a complete picture of E-Cardiology. These parameters/indicators may be comprised of temperature, blood pressure, oxygen saturation, heart rate, and ECG analysis. In addition, in differentiating between normal and abnormal heartbeats, a Smart CCU can also be used to detect QRS complexes in electrocardiographic (ECG) signals to determine the presence of a cardiac malady and different arrhythmias. Integrated and wearable IoT solutions, which address all the necessary cardiac parameters of a heart patient, need to be implemented. The results and accuracy of the devices/sensors used in the development of an IOT-based cardiac system cannot be compromised. The implemented system must be tested, evaluated, and approved under the supervision of cardiologists. Advanced communication technologies, including secure network protocols, must be implemented. Data accessibility features, such as widespread data access, must be possible in a secure environment so that the data confidentiality and integrity are maintained. Ubiquitous access is also an important factor and can be achieved by storing the digital data on a cloud server.

6.2. Limitations of the Review Study

This review of literature has some limitations. First, many papers were conference proceedings; therefore some parameters remained inaccessible since their authors did not mention them in detail. Second, some of the studies on AI-based IoT architecture in cardiac healthcare could not be located even after following a comprehensive search protocol, such as gray literature and reports that were not published in the databases which we selected for review. Therefore, we suggest that an additional systematic study be conducted to cover the related literature from other important databases.



Figure 6. Year-wise trends in publications in the field of E-Cardiology.

7. Conclusions

This review study outlines a total of 104 primary studies from 2016 to 2021 based on our filtering process for supporting the proposed research. Quality assessment of the selected studies was conducted for the formulated research questions after a rigorous analysis.

This work mentions different sensors and communication technologies being used in cardiology. Moreover, this review study also describes various preprocessing techniques and AI algorithms used in the existing studies to diagnose and classify CVD and ECG analysis. This systematic review also provides comparative analysis of various existing techniques in the field of AI, medical sensors, and communication technologies. Finally, this study targets various advantages and issues indicated in the existing literature in the field of E-Cardiology. The interaction of MIoT and artificial intelligence makes cardiac healthcare more manageable by making various applications, services, communication protocols, third party APIs, and IoT sensors available. E-Cardiology guarantees more privacy and security to the IoT devices which are prone to hackers. Furthermore, AI-based diagnosis of various cardiovascular diseases in E-Cardiology helps save time, enabling cardiologists to focus more on treatment. This systematic work presents a review protocol to analyse how IoT applications assist cardiac healthcare and how various artificial intelligence (AI) models contribute to present and prospective research work of IoT in E-Cardiology. This study also indicates how different communication technologies bring privacy and security to IoT devices related to cardiac healthcare. The purpose of this paper is to highlight the influence of IoT, communication technologies, and AI techniques in cardiac healthcare in light of our systematic literature review protocol. Therefore, one can say that this systematic review covers the complete and latest infrastructure of E-Cardiology, along with its benefits and challenges which have not been examined before in such a comprehensive way.

Author Contributions: Conceptualization, R.I. and M.A.K.; methodology, R.I. and U.U.; validation, U.U. and S.N.; formal analysis, U.U. and S.N.; investigation, U.U. and S.N.; resources, R.I., U.U. and S.N.; writing—original draft preparation, U.U., S.N. and A.U.; writing—review and editing, R.I., U.U. and S.N.; visualization, A.U. and S.N.; supervision, R.I. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Selected Literature

Summary of the selected literature consisting of 104 papers used in this systematic literature review is given in the Table A1.

Table A1. Selected Literature.

Sr.#	Ref#	Year	Paper Title
1	[17]	2016	Smart Real-Time Healthcare Monitoring and Tracking System using GSM/GPS Technologies
2	[58]	2016	Internet of Medical Things for Cardiac Monitoring: Paving The Way to 5G Mobile Networks
3	[59]	2016	IOT on Heart Attack Detection and Heart Rate Monitoring
4	[85]	2016	Heart Rate Monitoring System Using Finger Tip Through Arduino and Processing Software
5	[60]	2016	iCarMa: Inexpensive Cardiac Arrhythmia Management—An IoT Healthcare Analytics Solution
6	[94]	2016	Efficient Heart Disease Prediction System
7	[96]	2016	A new personalized ECG Signal Classification Algorithm using Block-based Neural Network and Particle Swarm Optimization
8	[97]	2016	Arrhythmia Recognition and Classification using Combined Linear and Nonlinear Features of ECG Signals
9	[98]	2016	Cardiac Arrhythmia Beat Classification Using DOST and PSO Tuned SVM
10	[99]	2016	High Performance Personalized Heartbeat Classification Model for Long-Term ECG Signal
11	[100]	2016	ECG Classification Using Wavelet Packet Entropy and Random Forests
12	[101]	2016	Automatic Coronary Artery Calcium Scoring in Cardiac CT Angiography using paired Convolutional Neural Networks
13	[61]	2016	ECG Signal Analysis and Arrhythmia Detection on IoT Wearable Medical Devices
14	[122]	2016	Study of IoT: Understanding IoT Architecture, Applications, Issues and Challenges
15	[121]	2016	Always Connected: The Security Challenges of the Healthcare Internet of Things
16	[136]	2016	Exploratory Study of Artificial Intelligence in Healthcare
17	[37]	2017	The IoT-based Heart Disease Monitoring System for Pervasive Healthcare Service
18	[62]	2017	Heartbeat Sensing and Heart Attack Detection using Internet of Things: IoT
19	[19]	2017	A Novel Cardiac Arrest Alerting System using IOT
20	[29]	2017	A Wearable Multiparameter Medical Monitoring And Alert System With First Aid
21	[63]	2017	Design And Implementation Of Low Cost Web Based Human Health Monitoring System Using Raspberry Pi 2
22	[64]	2017	Ultra-Low Power, Secure IoT Platform for Predicting Cardiovascular Diseases
23	[39]	2017	IOT Based Detection of Cardiac Arrythmia With Classification
24	[150]	2017	Student Research Abstract: A Novel IoT-based Wireless System to Monitor Heart Rate
25	[151]	2017	Cardiac Scan: A Non-contact and Continuous Heart-based User Authentication System
26	[102]	2017	Cardiac Arrhythmia Detection using Deep Learning
27	[103]	2017	Multiresolution Wavelet Transform based Feature Extraction and ECG Classification to Detect Cardiac Abnormalities
28	[104]	2017	ECG beat Classification using Empirical Mode Decomposition and Mixture of Features
29	[123]	2017	Internet of Medical Things (IOMT): Applications, Benefits and Future Challenges in Healthcare Domain
30	[152]	2018	Social Assistive Robot for Cardiac Rehabilitation: A Pilot Study with Patients with Angioplasty
31	[153]	2018	Impact of a Mobile Cycling Application on Cardiac Patients' Cycling Behavior and Enjoyment
32	[65]	2018	Pulse Oximetry and IOT based Cardiac Monitoring Integrated Alert System

Table A1. Cont.

Sr.#	Ref#	Year	Paper Title
33	[66]	2018	Detection of Cardiac Arrest Using Internet of Things
34	[18]	2018	Heart Attack Detection and Heart Rate Monitoring Using IoT
35	[38]	2018	IoT Based Continuous Monitoring of Cardiac Patients using Raspberry Pi
36	[20]	2018	Healthcare Monitoring System Based on Wireless Sensor Network for Cardiac Patients
37	[24]	2018	Heart Attack Detection By Heartbeat Sensing using Internet Of Things: IoT
38	[67]	2018	Real-Time Monitoring and Detection of "Heart Attack" Using Wireless Sensors and IoT
39	[105]	2018	Diagnosis of Shockable Rhythms for Automated External Defibrillators using a Reliable Support Vector Machine Classifier
40	[106]	2018	Automatic Recognition of Arrhythmia based on Principal Component Analysis Network and Linear Support Vector Machine
41	[107]	2018	Automated Recognition of Cardiac Arrhythmias using Sparse Decomposition over Composite Dictionary
42	[108]	2018	A Novel Adaptive Feature Extraction for Detection of Cardiac Arrhythmias using Hybrid technique MRDWT & MPNN Classifier from ECG Big Data
43	[109]	2018	Automated Diagnosis of Arrhythmia using Combination of CNN and LSTM Techniques with Variable Length Heart Beats
44	[11 0]	2018	A Deep Learning Approach for ECG-based Heartbeat Classification for Arrhythmia Detection
45	[111]	2018	A Novel Application of Deep Learning for Single-Lead ECG Classification
46	[91]	2018	Deep Learning for Prediction of Obstructive Disease From Fast Myocardial Perfusion SPECT A Multicenter Study
47	[124]	2018	Deploying Internet of Things in Healthcare: Benefits, Requirements, Challenges and Applications
48	[28]	2019	A Study on Heart Attack Detection by Heartbeat Monitoring Using IoT
49	[25]	2019	An Energy Efficient Wearable Smart IoT System to Predict Cardiac Arrest
50	[26]	2019	IoT Based Heart Attack Detection, Heart Rate and Temperature Monitor
51	[40]	2019	IoT based Diagnosing Myocardial Infarction through Firebase Web Application
52	[21]	2019	A Real-time Cardiac Monitoring using a Multisensory Smart IoT System
53	[68]	2019	An IoT based Efficient Hybrid Recommender System for Cardiovascular Disease
54	[69]	2019	Utilizing IoT Wearable Medical Device for Heart Disease Prediction using Higher Order Boltzmann Model: A Classification Approach
55	[30]	2019	The Cardiac Disease Predictor: IoT and ML Driven Healthcare System
56	[70]	2019	Machine Learning and IoT-based Cardiac Arrhythmia Diagnosis using Statistical and Dynamic Features of ECG
57	[71]	2019	Artificial Intelligence of Things Wearable System for Cardiac Disease Detection
58	[112]	2019	Neural Network Based Intelligent System for Predicting Heart Disease
59	[93]	2019	Cardiologist-level Arrhythmia Detection and Classification in ambulatory Electrocardiograms using a Deep Neural Network
60	[126]	2019	IoT Healthcare: Benefits, Issues and Challenges
61	[127]	2019	IoT, an Emerging Technology for Next Generation Medical Devices in Support of Cardiac Health Care—A Comprehensive Review
62	[11]	2019	IoT Technology, Applications and Challenges: A Contemporary Survey
63	[128]	2019	Internet of Things applications: A Systematic Review
64	[125]	2019	Smart Healthcare in the Era of Internet-of-Things
65	[130]	2019	Challenges and opportunities in IoT Healthcare Systems: A Systematic Review
66	[138]	2019	AI in Healthcare: Ethical and Privacy Challenges
67	[137]	2019	Key Challenges for Delivering Clinical Impact with Artificial Intelligence
68	[139]	2019	Healthcare uses of Artificial Intelligence: Challenges and Opportunities for Growth
69	[149]	2019	Artificial Intelligence in Clinical Decision Support: Challenges for Evaluating AI and Practical Implications
70	[72]	2020	An Efficient IoT-Based Platform for Remote Real-Time Cardiac Activity Monitoring
71	[86]	2020	IOT Based Heart Attack Detection & Heart Rate Monitoring System

Table A1. Cont.

Sr.#	Ref#	Year	Paper Title
72	[73]	2020	Remote Health and Monitoring, Heart Attack Detection and Location Tracking System with IoT
73	[74]	2020	IOT Based Heart Attack and Alcohol Detection in Smart Transportation and Accident
74	[75]	2020	HealthFog: An Ensemble Deep Learning based Smart Healthcare System for Automatic Diagnosis of Heart Diseases in Integrated IoT and Fog Computing Environments
75	[27]	2020	IoT based Health Care Monitoring Kit
76	[76]	2020	Automated Detection of Posterior Myocardial Infarction From VCG Signals Using Stationary Wavelet Transform Based Features
77	[22]	2020	IoT Based Real-Time Remote Patient Monitoring System
78	[77]	2020	Design, Fabrication, and Testing of an IoT Healthcare Cardiac Monitoring Device
79	[78]	2020	A Framework for Cardiac Arrhythmia Detection from IoT-based ECGs
80	[79]	2020	SAREF4health: Towards IoT Standard-based Ontology-Driven Cardiac E-health Systems
81	[23]	2020	An IoT Patient Monitoring Based on Fog Computing and Data Mining: Cardiac Arrhythmia Usecase
82	[113]	2020	CNN-KCL: Automatic Myocarditis Diagnosis using Convolutional Neural Network Combined with K-means Clustering
83	[114]	2020	Detection of Myocardial Infarction Based on Novel Deep Transfer Learning Methods for Urban Healthcare in Smart Cities
84	[115]	2020	Automatic Diagnosis of the 12-lead ECG using a Deep Neural Network
85	[129]	2020	Internet of Things Based Distributed Healthcare Systems: A Review
86	[131]	2020	IoT-Enabled Healthcare: Benefits, Issues and Challenges
87	[132]	2020	A Comprehensive Review on the Emerging IoT Cloud based Technologies for Smart Healthcare
88	[140]	2020	Ethical Challenges of Integrating AI into Healthcare
89	[80]	2021	Monitoring Patients to Prevent Myocardial Infarction using Internet of Things Technology
90	[81]	2021	Filtering the ECG Signal towards Heart Attack Detection using Motion Artifact Removal Technique
91	[82]	2021	Artificial-Intelligence-Enhanced Mobile System for Cardiovascular Health Management
92	[83]	2021	AMBtalk: A Cardiovascular IoT Device for Ambulance Applications
93	[84]	2021	Predicting Cardiovascular Events with Deep Learning Approach in the Context of the Internet of Things
94	[41]	2021	IoT Based Wearable Monitoring Structure for detecting Abnormal Heart
95	[154]	2021	An Advanced Patient Health Monitoring System
96	[155]	2021	Development of Smart Health Monitoring System using Internet of Things
97	[116]	2021	Early Detection of Myocardial Infarction in Low-Quality Echocardiography
98	[119]	2021	Prediction of Heart Disease Using Deep Convolutional Neural Networks
99	[117]	2021	AI-Based Smart Prediction of Clinical Disease Using Random Forest Classifier and Naive Bayes
100	[118]	2021	Artificial Intelligence Enhanced Electrocardiogram for the Early Detection of Cardiac Amyloidosis
101	[133]	2021	IOT in Healthcare: Challenges, Benefits, Applications and Opportunities
102	[134]	2021	A Survey on IoT Smart Healthcare: Emerging Technologies, Applications, Challenges, and Future Trends
103	[141]	2021	Secure and Robust Machine Learning for Healthcare: A Survey
104	[142]	2021	AI in Healthcare: Medical and Socio-Economic Benefits and Challenges

References

- 1. WHO: The Top 10 Causes of Death. Available online: http://www.who.int/mediacentre/factsheets/fs310/en/index.html (accessed on 10 April 2020).
- 2. Available online: https://www.who.int/news-room/fact-sheets/detail/cardiovasculardiseases-(cvds) (accessed on 11 November 2020).
- Umar, U.; Khan, M.A.; Irfan, R.; Ahmad, J. IoT-based Cardiac Healthcare System for Ubiquitous Healthcare Service. In Proceedings of the 2021 International Congress of Advanced Technology and Engineering (ICOTEN), Taiz, Yemen, 4–5 July 2021; IEEE: New York, NY, USA, 2021; pp. 1–6.
- 4. Internet of Things: Internet of Things. Available online: https://www.en.wikipedia.ord/wiki/internetofthings (accessed on 10 April 2020).

- Bao, J.; Shou, X.; Wang, H.; Yang, H. Study on heartbeat information acquired from pressure cushion based on body sensor network. In Proceedings of the 2013 IEEE International Conference on Green Computing and Communications and IEEE Internet of Things and IEEE Cyber, Physical and Social Computing, Beijing, China, 20–23 August 2013; pp. 1103–1108.
- 6. Lin, C.-T.; Chuang, C.-H.; Huang, C.-S.; Tsai, S.-F.; Lu, S.-W.; Chen, Y.-H.; Ko, L.-W. Wireless and wearable eeg system for evaluating driver vigilance. *IEEE Trans. Biomed. Circuits Syst.* **2014**, *8*, 165–176. [PubMed]
- 7. Benhar, H.; Idri, A.; Fernandez-Aleman, J. Data preprocessing for heart disease classification: A systematic literature review. *Comput. Methods Programs Biomed.* 2020, 195, 105635. [CrossRef] [PubMed]
- 8. Shah, R.; Chircu, A. Iot and ai in healthcare: A systematic literature review. Issues Inf. Syst. 2018, 19, 33–41.
- Kadi, I.; Idri, A.; Fernandez-Aleman, J. Knowledge discovery in cardiology: A systematic literature review. *Int. J. Med. Inform.* 2017, 97, 12–32. [CrossRef] [PubMed]
- Shah, N.; Ali, Y.; Ullah, N.; García-Magariño, I. Internet of things for healthcare using effects of mobile computing: A systematic literature review. *Wirel. Commun. Mob. Comput.* 2019, 2019, 5931315.
- Balaji, S.; Nathani, K.; Santhakumar, R. Iot technology, applications and challenges: A contemporary survey. *Wirel. Pers. Commun.* 2019, 108, 363–388. [CrossRef]
- AbdElnapi, N.M.M.; Omran, N.F.; Ali, A.A.; Omara, F.A. A survey of internet of things technologies and projects for healthcare services. In Proceedings of the 2018 International Conference on Innovative Trends in Computer Engineering (ITCE), Aswan, Egypt, 19–21 February 2018; pp. 48–55.
- Al-rawashdeh, M.; Keikhosrokiani, P.; Belaton, B.; Alawida, M.; Zwiri, A. IoT Adoption and Application for Smart Healthcare: A Systematic Review. Sensors 2022, 22, 5377. [CrossRef]
- 14. Kashani, O.H.; Madanipour, M.; Nikravan, M.; Asghari, P.; Mahdipour, E. A systematic review of IoT in healthcare: Applications, techniques, and trends. *J. Netw. Comput. Appl.* **2021**, *192*, 103164. [CrossRef]
- 15. Esfandiari, N.; Babavalian, M.R.; Moghadam, A.-M.E.; Tabar, V.K. Knowledge discovery in medicine: Current issue and future trend. *Expert Syst. Appl.* **2014**, *41*, 4434–4463. [CrossRef]
- Wohlin, C. Guidelines for snowballing in systematic literature studies and a replication in software engineering. In Proceedings
 of the 18th International Conference on Evaluation and Assessment in Software Engineering, London, UK, 13–14 May 2014;
 pp. 1–10.
- Aziz, K.; Tarapiah, S.; Ismail, S.H.; Atalla, S. Smart real-time healthcare monitoring and tracking system using gsm/gps technologies. In Proceedings of the 2016 3rd MEC International Conference on Big Data and Smart City (ICBDSC), Muscat, Oman, 15–16 March 2016; pp. 1–7.
- Patel, N.; Patel, P.; Patel, N. Heart attack detection and heart rate monitoring using iot. Int. J. Innov. Adv. Comput. Sci. (IJIACS) 2018, 7, 611–615.
- 19. Ajay, H.; Rao, A.; Balavanan, M.; Lalit, R. A novel cardiac arrest alerting system using iot. Int. J. Sci. Technol. Eng. 2017, 3, 78-83.
- Gogate, U.; Bakal, J. Healthcare monitoring system based on wireless sensor network for cardiac patients. *Biomed. Pharmacol.* 2018, 11, 1681. [CrossRef]
- Majumder, A.J.; Elsaadany, M.; Izaguirre, J.A.; Ucci, D.R. A realtime cardiac monitoring using a multisensory smart iot system. In Proceedings of the 2019 IEEE 43rd Annual Computer Software and Applications Conference (COMPSAC), Milwaukee, WI, USA, 15–19 July 2019; Volume 2, pp. 281–287.
- Yew, H.T.; Ng, M.F.; Ping, S.Z.; Chung, S.K.; Chekima, A.; Dargham, J.A. Iot based real-time remote patient monitoring system. In Proceedings of the 2020 16th IEEE International Colloquium on Signal Processing & Its Applications (CSPA), Langkawi Island, Malaysia, 28–29 February 2020; pp. 176–179.
- 23. Moghadas, E.; Rezazadeh, J.; Farahbakhsh, R. An iot patient monitoring based on fog computing and data mining: Cardiac arrhythmia usecase. *Internet Things* **2020**, *11*, 100251. [CrossRef]
- 24. Gurjar, A.; Sarnaik, N.A. Heart attack detection by heartbeat sensing using Internet of Things: IoT. *Int. Res. J. Eng. Technol.* **2018**, *5*, 3332–3335.
- 25. Majumder, A.; ElSaadany, Y.A.; Young, R.; Ucci, D.R. An energy efficient wearable smart iot system to predict cardiac arrest. *Adv.-Hum. Interact.* **2019**, 2019, 1507465. [CrossRef]
- Vaishnave, A.K.; Jenisha, S.T.; Tamil Selvi, S. IoT Based Heart Attack Detection, Heart Rate and Temperature Monitor. Int. Res. J. Multidiscip. Technovat. 2019, 1, 61–70.
- Acharya, A.D.; Patil, S.N. Iot based health care monitoring kit. In Proceedings of the 2020 Fourth International Conference on Computing Methodologies and Communication (ICCMC), Erode, India, 11–13 March 2020; pp. 363–368.
- Shivakumar, E.; Hiremani, N. A Study on Heart Attack Detection by Heartbeat Monitoring Using Iot. Int. Res. J. Eng. Technol. 2019, 6, 4016–4020.
- 29. Manimaraboopathy, M.; Vijayalakshmi, S.; Hemavathy, D.; Priya, A. A wearable multiparameter medical monitoring and alert system with first aid. *Int. J. Smart Sens. Intell. Syst.* 2017, 10, 446. [CrossRef]
- Tabassum, S.; Zaman, M.I.U.; Ullah, M.S.; Rahaman, A.; Nahar, S.; Islam, A.M. The cardiac disease predictor: Iot and ml driven healthcare system. In Proceedings of the 2019 4th International Conference on Electrical Information and Communication Technology (EICT), Khulna, Bangladesh, 20–22 December 2019; pp. 1–6.
- 31. Available online: https://bit.ly/3Eo4IEG (accessed on 11 September 2021).
- 32. Available online: https://bit.ly/3nALDTK (accessed on 11 September 2021).

- 33. Available online: https://datasheetspdf.com/datasheet/KY-039.html (accessed on 11 September 2021).
- 34. Available online: https://iss.jaxa.jp/en/kiboexp/pm/holter/holter.pdf (accessed on 11 September 2021).
- Available online: https://www.amazon.com/oximeter-fingertip-saturation-batteries-GreyWhite/dp/B081JQJVPD (accessed on 11 September 2021).
- 36. Available online: https://datasheets.maximintegrated.com/en/ds/MAX30100.pdf (accessed on 11 September 2021).
- Li, C.; Hu, X.; Zhang, L. The iot-based heart disease monitoring system for pervasive healthcare service. *Procedia Comput. Sci.* 2017, 112, 2328–2334. [CrossRef]
- Mathivanan, M.; Balamurugan, M.; Nandini; Reddy, M. Iot based continuous monitoring of cardiac patients using raspberry pi. AIP Conf. Proc. 2018, 2039, 020025.
- 39. Shah, J.; Danve, S. Iot Based Detection of Cardiac Arrythmia with Classification. In Proceedings of the International Conference on Emerging Trends in Engineering, Technology, Science and Management (ICETETSM-17), Delhi, India, 11 June 2017.
- Sharma, A.K.; Saini, L.M. Iot based diagnosing myocardial infarction through firebase web application. In Proceedings of the 2019 3rd International Conference on Electronics, Communication and Aerospace Technology (ICECA), Coimbatore, India, 12–14 June 2019; pp. 190–195.
- Kora, P.; Rajani, A.; Chinnaiah, M.; Swaraja, K.; Meenakshi, K. Iot based wearable monitoring structure for detecting abnormal heart. In Proceedings of the 2021 International Conference on Sustainable Energy and Future Electric Transportation (SEFET), Hyderabad, India, 21–23 January 2021; pp. 1–4.
- 42. Available online: https://pdf1.alldatasheet.com/datasheetpdf/view/517588/TI1/LM35.html (accessed on 11 September 2021).
- 43. Available online: https://www.alldatasheet.com/datasheetpdf/pdf/58557/DALLAS/DS18B20.html (accessed on 11 September 2021).
- 44. Available online: https://pdf1.alldatasheet.com/datasheetpdf/view/109863/MICROCHIP/MCP9700.html (accessed on 11 September 2021).
- 45. Available online: https://www.ti.com/lit/gpn/tmp100 (accessed on 11 September 2021).
- 46. Available online: https://www.nxp.com/docs/en/data-sheet/MPX10.pdf (accessed on 11 September 2021).
- 47. Available online: https://bit.ly/3GrPBXl (accessed on 11 September 2021).
- 48. Available online: https://www.hospitalsstore.com/omron-hbp-1300-digital-automaticblood-pressure-monitor/ (accessed on 11 September 2021).
- 49. Available online: https://www.vernier.com/product/blood-pressure-sensor/ (accessed on 11 September 2021).
- 50. Available online: https://bit.ly/3CgUsIm (accessed on 11 September 2021).
- 51. Available online: https://bit.ly/3Gq1YmP (accessed on 11 September 2021).
- 52. Available online: https://www.alldatasheet.com/datasheetpdf/pdf/1007828/AD/ADAS1000-1BCPZ.html (accessed on 11 September 2021).
- 53. Available online: https://pdf1.alldatasheet.com/datasheetpdf/view/902688/AD/AD8233.html (accessed on 11 September 2021).
- 54. Available online: https://www.shimmersensing.com/products/shimmer3-ecgsensorspecifications-tab (accessed on 11 September 2021).
- 55. Available online: http://www.shimmersensing.com/products/ecg-development-kit (accessed on 11 September 2021).
- 56. Available online: https://bit.ly/2ZujQvT (accessed on 11 September 2021).
- 57. Available online: http://www.shimmersensing.com/products/ecg-developmentkitdownload-tab (accessed on 11 September 2021).
- Jusak, J.; Pratikno, H.; Putra, V.H. Internet of medical things for cardiac monitoring: Paving the way to 5 g mobile networks. In Proceedings of the 2016 IEEE International Conference on Communication, Networks and Satellite (COMNETSAT), Surabaya, Indonesia, 8–10 December 2016; pp. 75–79.
- 59. Manisha, M.; Neeraja, K.; Sindhura, V.; Ramaya, P. Iot on heart attack detection and heart rate monitoring. *Int. J. Innov. Eng. Technol.* (*IJIET*) **2016**, 196.
- Puri, C.; Ukil, A.; Bandyopadhyay, S.; Singh, R.; Pal, A.; Mandana, K. Icarma: Inexpensive cardiac arrhythmia management—An iot healthcare analytics solution. In Proceedings of the First Workshop on IoT-Enabled Healthcare and Wellness Technologies and Systems, Singapore, 30 June 2016; pp. 3–8.
- Azariadi, D.; Tsoutsouras, V.; Xydis, S.; Soudris, D. Ecg signal analysis and arrhythmia detection on iot wearable medical devices. In Proceedings of the 2016 5th International Conference on Modern Circuits and Systems Technologies (MOCAST), Thessaloniki, Greece, 12–14 May 2016; pp. 1–4.
- 62. Sidheeque, A.; Kumar, A.; Balamurugan, R.; Deepak, K.; Sathish, K. Heartbeat sensing and heart attack detection using internet of things: Iot. *Int. J. Eng. Sci. Comput.* **2017**, *7*, 6662–6666.
- Kirankumar, C.; Prabhakaran, M. Design and implementation of low cost web based human health monitoring system using raspberry pi 2. In Proceedings of the 2017 IEEE International Conference on Electrical, Instrumentation and Communication Engineering (ICEICE), Karur, India, 27–28 April 2017; pp. 1–5.
- 64. Yasin, M.; Tekeste, T.; Saleh, H.; Mohammad, B.; Sinanoglu, O.; Ismail, M. Ultra-low power, secure iot platform for predicting cardiovascular diseases. *IEEE Trans. Circuits Syst. I Regul. Pap.* **2017**, *64*, 2624–2637. [CrossRef]
- Murali, D.; Rao, D.R.; Rao, S.R.; Ananda, M. Pulse oximetry and iot based cardiac monitoring integrated alert system. In Proceedings of the 2018 International Conference on Advances in Computing, Communications and Informatics (ICACCI), Bengaluru, India, 19–22 September 2018; pp. 2237–2243.

- 66. Hemalatha, V.; Poykasi, K.; Sangeetha, P.; Gomathi, P. Detection of cardiac arrest using internet of things. *Int. J. Sci. Res. Comput. Sci.* 2018, *3*, 793–801.
- Geethalakshmi, T.S.T. Real-time monitoring and detection of "heart attack" using wireless sensors and iot. In Proceedings of the 2010 Fourth International Conference on Sensor Technologies and Applications, Venice, Italy, 18–25 July 2010.
- 68. Jabeen, F.; Maqsood, M.; Ghazanfar, M.A.; Aadil, F.; Khan, S.; Khan, M.F.; Mehmood, I. An iot based efficient hybrid recommender system for cardiovascular disease. *Peer-Netw. Appl.* **2019**, *12*, 1263–1276. [CrossRef]
- 69. Al-Makhadmeh, Z.; Tolba, A. Utilizing iot wearable medical device for heart disease prediction using higher order boltzmann model: A classification approach. *Measurement* **2019**, *147*, 106815. [CrossRef]
- Devi, R.L.; Kalaivani, V. Machine learning and iot-based cardiac arrhythmia diagnosis using statistical and dynamic features of ecg. J. Supercomput. 2020, 76, 6533–6544. [CrossRef]
- Lin, Y.-J.; Chuang, C.-W.; Yen, C.-Y.; Huang, S.-H.; Huang, P.-W.; Chen, J.-Y.; Lee, S.-Y. Artificial intelligence of things wearable system for cardiac disease detection. In Proceedings of the 2019 IEEE International Conference on Artificial Intelligence Circuits and Systems (AICAS), Hsinchu, Taiwan, 18–20 March 2019; pp. 67–70.
- 72. Raj, S. An efficient iot-based platform for remote real-time cardiac activity monitoring. *IEEE Trans. Consum. Electron.* 2020, 66, 106–114. [CrossRef]
- 73. Lokesh, S.; Pon Nandhakuamr, S.; Ranjith, R. Remote health and monitoring, heart attack detection and location tracking system with iot. *Hindusthan J. Inf. Commun. Mod. Comput.* **2020**, *1*, 1–5.
- Kumari, K.V.R.; Sarma, G. Iot Based Heart Attack and Alcohol Detection in Smart Transportation and Accident Prevention for Vehicle Drivers. Int. J. Recent Dev. Sci. Technol. 2020, 4, 179–185.
- Tuli, S.; Basumatary, N.; Gill, S.S.; Kahani, M.; Arya, R.C.; Wander, G.S.; Buyya, R. Healthfog: An ensemble deep learning based smart healthcare system for automatic diagnosis of heart diseases in integrated iot and fog computing environments. *Future Gener. Syst.* 2020, 104, 187–200. [CrossRef]
- 76. Prabhakararao, E.; Dandapat, S. Automated detection of posterior myocardial infarction from vcg signals using stationary wavelet transform based features. *IEEE Sens. Lett.* **2020**, *4*, 1–4. [CrossRef]
- Zagan, I.; Gaitan, V.G.; Petrariu, A.-I.; Iuga, N.; Brezulianu, A. Design, fabrication, and testing of an iot healthcare cardiac monitoring device. *Computers* 2020, 9, 15. [CrossRef]
- He, J.; Rong, J.; Sun, L.; Wang, H.; Zhang, Y.; Ma, J. A framework for cardiac arrhythmia detection from iot-based ecgs. World Wide Web 2020, 23, 2835–2850. [CrossRef]
- 79. Moreira, J.; Pires, L.F.; van Sinderen, M.; Daniele, L.; GirodGenet, M. Saref4health: Towards iot standard-based ontology-driven cardiac e-health systems. *Appl. Ontol.* **2020**, *15*, 385–410. [CrossRef]
- 80. Akhoondan, F.; Hamidi, H.; Broumandnia, A. Monitoring patients to prevent myocardial infarction using internet of things technology. J. Community Health Res. 2021, 10, 52–59. [CrossRef]
- Selvaraj, S.; Ramya, P.; Priya, R.; Ramya, C. Filtering the ecg signal towards heart attack detection using motion artifact removal technique. In Proceedings of the 2021 Third International Conference on Intelligent Communication Technologies and Virtual Mobile Networks (ICICV), Tirunelveli, India, 4–6 February 2021; pp. 185–188.
- 82. Fu, Z.; Hong, S.; Zhang, R.; Du, S. Artificial-intelligence-enhanced mobile system for cardiovascular health management. *Sensors* 2021, 21, 773. [CrossRef]
- 83. Chen, W.-L.; Lin, Y.-B.; Chang, T.C.-Y.; Lin, Y.-R. Ambtalk: A cardiovascular iot device for ambulance applications. *Sensors* **2021**, 21, 2781. [CrossRef] [PubMed]
- 84. Dami, S.; Yahaghizadeh, M. Predicting cardiovascular events with deep learning approach in the context of the internet of things. *Neural Comput. Appl.* **2021**, *33*, 7979–7996. [CrossRef]
- 85. Mallick, B.; Patro, A.K. Heart rate monitoring system using finger tip through arduino and processing software. *Int. J. Sci. Eng. Technol. Res.* (*IJSETR*) **2016**, *5*, 84–89.
- Giri, S.; Kumar, U.; Sharma, V.; Kumar, S.; Kumari, S.; Rawani, R.K.; Pande, P.; Bhadra, B. Iot based heart attack detection & heart rate monitoring system. In Proceedings of the International Conference on Recent Trends in Artificial Intelligence, Iot, Smart Cities & Application (ICAISC 2020), Jharkhand, India, 16 July 2020.
- 87. Nygards, M.-E.; Hulting, J. An automated system for ecg monitoring. Comput. Biomed. Res. 1979, 12, 181–202. [CrossRef]
- 88. Romiti, S.; Vinciguerra, M.; Saade, W.; Cortajarena, I.A.; Greco, E. Artificial intelligence (AI) and cardiovascular diseases: An unexpected alliance. *Cardiol. Res. Pract.* 2020, 2020, 4972346. [CrossRef]
- 89. Alsharqi, M.; Woodward, W.; Mumith, J.; Markham, D.; Upton, R.; Leeson, P. Artificial intelligence and echocardiography. *Echo Res. Pract.* **2018**, *5*, R115–R125. [CrossRef] [PubMed]
- Samad, M.D.; Ulloa, A.; Wehner, G.J.; Jing, L.; Hartzel, D.; Good, C.W.; Williams, B.A.; Haggerty, C.M.; Fornwalt, B.K. Predicting survival from large echocardiography and electronic health record datasets: Optimization with machine learning. *JACC Cardiovasc. Imaging* 2019, 12, 681–689. [CrossRef] [PubMed]
- Betancur, J.; Commandeur, F.; Motlagh, M.; Sharir, T.; Einstein, A.J.; Bokhari, S.; Fish, M.B.; Ruddy, T.D.; Kaufmann, P.; Sinusas, A.J.; et al. Deep learning for prediction of obstructive disease from fast myocardial perfusion spect: A multicenter study. *JACC Cardiovasc. Imaging* 2018, *11*, 1654–1663. [CrossRef] [PubMed]
- Seetharam, K.; Kagiyama, N.; Sengupta, P.P. Application of mobile health, telemedicine and artificial intelligence to echocardiography. Echo Res. Pract. 2019, 6, R41–R52. [CrossRef] [PubMed]

- Hannun, A.Y.; Rajpurkar, P.; Haghpanahi, M.; Tison, G.H.; Bourn, C.; Turakhia, M.P.; Ng, A.Y. Cardiologist-level arrhythmia detection and classification in ambulatory electrocardiograms using a deep neural network. *Nat. Med.* 2019, 25, 65–69. [CrossRef]
 Saxena, K.; Sharma, R. Efficient heart disease prediction system. *Procedia Comput. Sci.* 2016, 85, 962–969.
- 95. Learning, M. Heart disease diagnosis and prediction using machine learning and data mining techniques: A review. *Adv. Comput. Sci. Technol.* **2017**, *10*, 2137–2159.
- Shadmand, S.; Mashoufi, B. A new personalized ecg signal classification algorithm using block-based neural network and particle swarm optimization. *Biomed. Signal Process. Control* 2016, 25, 12–23. [CrossRef]
- 97. Elhaj, F.A.; Salim, N.; Harris, A.R.; Swee, T.T.; Ahmed, T. Arrhythmia recognition and classification using combined linear and nonlinear features of ecg signals. *Comput. Methods Programs Biomed.* **2016**, *127*, 52–63. [CrossRef]
- 98. Raj, S.; Ray, K.C.; Shankar, O. Cardiac arrhythmia beat classification using dost and pso tuned svm. *Comput. Methods Programs Biomed.* **2016**, 136, 163–177. [CrossRef] [PubMed]
- 99. Li, P.; Wang, Y.; He, J.; Wang, L.; Tian, Y.; Zhou, T.-S.; Li, T.; Li, J.-S. High-performance personalized heartbeat classification model for long-term ecg signal. *IEEE Trans. Biomed. Eng.* **2016**, *64*, 78–86. [PubMed]
- 100. Li, T.; Zhou, M. Ecg classification using wavelet packet entropy and random forests. Entropy 2016, 18, 285. [CrossRef]
- 101. Wolterink, J.M.; Leiner, T.; de Vos, B.D.; van Hamersvelt, R.W.; Viergever, M.A.; Isgum, I. Automatic coronary artery calcium scoring in cardiac ct angiography using paired convolutional neural networks. *Med. Image Anal.* 2016, 34, 123–136. [CrossRef] [PubMed]
- 102. Isin, A.; Ozdalili, S. Cardiac arrhythmia detection using deep learning. Procedia Comput. Sci. 2017, 120, 268–275. [CrossRef]
- Sahoo, S.; Kanungo, B.; Behera, S.; Sabut, S. Multiresolution wavelet transform based feature extraction and ecg classification to detect cardiac abnormalities. *Measurement* 2017, 108, 55–66. [CrossRef]
- 104. Sahoo, S.; Mohanty, M.; Behera, S.; Sabut, S.K. Ecg beat classification using empirical mode decomposition and mixture of features. *J. Med. Eng. Technol.* **2017**, *41*, 652–661. [CrossRef]
- Nguyen, M.T.; Shahzad, A.; van Nguyen, B.; Kim, K. Diagnosis of shockable rhythms for automated external defibrillators using a reliable support vector machine classifier. *Biomed. Signal Process. Control* 2018, 44, 258–269. [CrossRef]
- 106. Yang, W.; Si, Y.; Wang, D.; Guo, B. Automatic recognition of arrhythmia based on principal component analysis network and linear support vector machine. *Comput. Biol. Med.* **2018**, 101, 22–32. [CrossRef]
- Raj, S.; Ray, K.C. Automated recognition of cardiac arrhythmias using sparse decomposition over composite dictionary. *Comput. Methods Programs Biomed.* 2018, 165, 175–186. [CrossRef] [PubMed]
- Rai, H.M.; Chatterjee, K. A novel adaptive feature extraction for detection of cardiac arrhythmias using hybrid technique mrdwt & mpnn classifier from ecg big data. *Big Data Res.* 2018, 12, 13–22.
- Oh, S.L.; Ng, E.Y.; Tan, R.S.; Acharya, U.R. Automated diagnosis of arrhythmia using combination of cnn and lstm techniques with variable length heart beats. *Comput. Biol. Med.* 2018, 102, 278–287. [CrossRef] [PubMed]
- 110. Sannino, G.; de Pietro, G. A deep learning approach for ecg-based heartbeat classification for arrhythmia detection. *Future Gener. Comput. Syst.* **2018**, *86*, 446–455. [CrossRef]
- Mathews, S.M.; Kambhamettu, C.; Barner, K.E. A novel application of deep learning for single-lead ecg classification. *Comput. Biol. Med.* 2018, 99, 53–62. [CrossRef]
- 112. Subhadra, K.; Vikas, B. Neural network based intelligent system for predicting heart disease. *Int. J. Innov. Technol. Explor. Eng.* **2019**, *8*, 484–487.
- 113. Sharifrazi, D.; Alizadehsani, R.; Joloudari, J.H.; Shamshirband, S.; Hussain, S.; Sani, Z.A.; Hasanzadeh, F.; Shoaibi, A.; Dehzangi, A.; Alinejad-Rokny, H. Cnn-kcl: Automatic myocarditis diagnosis using convolutional neural network combined with k-means clustering. *Preprints* 2020, 2020070650. [CrossRef]
- Alghamdi, A.; Hammad, M.; Ugail, H.; Abdel-Raheem, A.; Muhammad, K.; Khalifa, H.S.; El-Latif, A.; Ahmed, A. Detection of myocardial infarction based on novel deep transfer learning methods for urban healthcare in smart cities. *Multimed. Tools Appl.* 2020, 1–22. [CrossRef]
- 115. Ribeiro, A.H.; Ribeiro, M.H.; Paixao, G.M.; Oliveira, D.M.; Gomes, P.R.; Canazart, J.A.; Ferreira, M.P.; Andersson, C.R.; Macfarlane, P.W.; Meira, W., Jr.; et al. Automatic diagnosis of the 12-lead ecg using a deep neural network. *Nat. Commun.* 2020, *11*, 1760. [CrossRef]
- Degerli, A.; Zabihi, M.; Kiranyaz, S.; Hamid, T.; Mazhar, R.; Hamila, R.; Gabbouj, M. Early detection of myocardial infarction in lowquality echocardiography. *IEEE Access* 2021, *9*, 34442–34453. [CrossRef]
- 117. Jackins, V.; Vimal, S.; Kaliappan, M.; Lee, M.Y. Ai-based smart prediction of clinical disease using random forest classifier and naive bayes. *J. Supercomput.* 2021, 77, 5198–5219. [CrossRef]
- 118. Grogan, M.; Lopez-Jimenez, F.; Cohen-Shelly, M.; Dispenzieri, A.; Attia, Z.I.; Ezzedine, O.F.A.; Lin, G.; Kapa, S.; Borgeson, D.D.; Friedman, P.A.; et al. Artificial intelligence-enhanced electrocardiogram for the early detection of cardiac amyloidosis. In *Mayo Clinic Proceedings*; Elsevier: Amsterdam, The Netherlands, 2021.
- Mehmood, A.; Iqbal, M.; Mehmood, Z.; Irtaza, A.; Nawaz, M.; Nazir, T.; Masood, M. Prediction of heart disease using deep convolutional neural networks. *Arab. J. Sci. Eng.* 2021, 46, 3409–3422. [CrossRef]
- 120. Available online: https://www.upgrad.com/blog/data-preprocessing-in-machine-learning (accessed on 25 December 2021).
- Williams, P.A.; McCauley, V. Always connected: The security challenges of the healthcare internet of things. In Proceedings of the 2016 IEEE 3rd World Forum on Internet of Things (WF-IoT), Reston, VA, USA, 12–14 December 2016; pp. 30–35.

- 122. Soumyalatha, S.G.H. Study of iot: Understanding iot architecture, applications, issues and challenges. In Proceedings of the 1st International Conference on Innovations in Computing & Networking (ICICN16), Bengaluru, India, 12–13 May 2016.
- Joyia, G.J.; Liaqat, R.M.; Farooq, A.; Rehman, S. Internet of medical things (IOMT): Applications, benefits and future challenges in healthcare domain. J. Commun. 2017, 12, 240–247. [CrossRef]
- 124. Shehabat, I.M.; Al-Hussein, N. Deploying internet of things in healthcare: Benefits, requirements, challenges and applications. *J. Commun.* **2018**, *13*, 574–580. [CrossRef]
- 125. Zhu, H.; Wu, C.K.; Koo, C.H.; Tsang, Y.T.; Liu, Y.; Chi, H.R.; Tsang, K.-F. Smart healthcare in the era of internet-of-things. *IEEE Consum. Electron. Mag.* 2019, *8*, 26–30. [CrossRef]
- 126. de Michele, R.; Furini, M. Iot healthcare: Benefits, issues and challenges. In Proceedings of the 5th EAI International Conference on Smart Objects and Technologies for Social Good, Valencia, Spain, 25–27 September 2019; pp. 160–164.
- 127. Pavithr, K.; Saravanan, G. Iot, an emerging technology for next generation medical devices in support of cardiac health care—A comprehensive review. *Int. Res. J. Multidiscip. Technovat.* **2019**, *1*, 35–40. [CrossRef]
- 128. Asghari, P.; Rahmani, A.M.; Javadi, H.H.S. Internet of things applications: A systematic review. *Comput. Netw.* **2019**, *148*, 241–261. [CrossRef]
- 129. Birje, M.N.; Hanji, S.S. Internet of things based distributed healthcare systems: A review. J. Data Inf. Manag. 2020, 2, 149–165. [CrossRef]
- 130. Selvaraj, S.; Sundaravaradhan, S. Challenges and opportunities in iot healthcare systems: A systematic review. *SN Appl. Sci.* **2020**, 2, 1–8. [CrossRef]
- Al-Shargabi, B.; Abuarqoub, S. Iot-enabled healthcare: Benefits, issues and challenges. In Proceedings of the 4th International Conference on Future Networks and Distributed Systems (ICFNDS), St. Petersburg, Russia, 26–27 November 2020; pp. 1–5. [CrossRef]
- Isravel, D.P.; Silas, S. A comprehensive review on the emerging iot-cloud based technologies for smart healthcare. In Proceedings of the 2020 6th International Conference on Advanced Computing and Communication Systems (ICACCS), Coimbatore, India, 6–7 March 2020; pp. 606–611.
- Trayush, T.; Bathla, R.; Saini, S.; Shukla, V.K. Iot in healthcare: Challenges, benefits, applications, and opportunities. In Proceedings of the 2021 International Conference on Advance Computing and Innovative Technologies in Engineering (ICACITE), Greater Noida, India, 30–31 December 2021; pp. 107–111.
- 134. Tunc, M.A.; Gures, E.; Shayea, I. A survey on iot smart healthcare: Emerging technologies, applications, challenges, and future trends. *arXiv* 2021, arXiv:2109.02042.
- 135. Tsoutsouras, V.; Azariadi, D.; Koliogewrgi, K.; Xydis, S.; Soudris, D. Software design and optimization of ecg signal analysis and diagnosis for embedded iot devices. In *Components and Services for IoT Platforms*; Springer International Publishing: Cham, Switzerland, 2017; pp. 299–322.
- 136. Alugubelli, R. Exploratory study of artificial intelligence in healthcare. Int. J. Innov. Eng. Res. Technol. 2016, 3, 1–10.
- Kelly, C.J.; Karthikesalingam, A.; Suleyman, M.; Corrado, G.; King, D. Key challenges for delivering clinical impact with artificial intelligence. *BMC Med.* 2019, 17, 1–9. [CrossRef]
- 138. Bartoletti, I. Ai in healthcare: Ethical and privacy challenges. In Proceedings of the Conference on Artificial Intelligence in Medicine in Europe, Poznan, Poland, 26–29 June 2019; Springer International Publishing: Cham, Switzerland, 2019; pp. 7–10.
- Racine, E.; Boehlen, W.; Sample, M. Healthcare uses of artificial intelligence: Challenges and opportunities for growth. In *Healthcare Management Forum*; SAGE Publications: Los Angeles, CA, USA, 2019; Volume 32, pp. 272–275.
- 140. Lehmann, L.S. Ethical challenges of integrating ai into healthcare. In *Artificial Intelligence in Medicine;* Springer International Publishing: Cham, Switzerland, 2020; pp. 1–5.
- 141. Qayyum, A.; Qadir, J.; Bilal, M.; Al-Fuqaha, A. Secure and robust machine learning for healthcare: A survey. *IEEE Rev. Biomed. Eng.* **2020**, *14*, 156–180. [CrossRef] [PubMed]
- 142. Shaheen, M.Y. AI in Healthcare: Medical and Socio-Economic Benefits and Challenges. Sci. Prepr. 2021. [CrossRef]
- Feeny, A.K.; Chung, M.K.; Madabhushi, A.; Attia, Z.I.; Cikes, M.; Firouznia, M.; Friedman, P.A.; Kalscheur, M.M.; Kapa, S.; Narayan, S.M.; et al. Artificial intelligence and machine learning in arrhythmias and cardiac electrophysiology. *Circ. Arrhythmia Electrophysiol.* 2020, 13, 007952. [CrossRef] [PubMed]
- 144. Hammoudeh, M.; Epiphaniou, G.; Belguith, S.; Unal, D.; Adebisi, B.; Baker, T.; Kayes, A.; Watters, P. A service-oriented approach for sensing in the internet of things: Intelligent transportation systems and privacy use cases. *IEEE Sens. J.* 2020, 21, 15753–15761. [CrossRef]
- 145. Abuarqoub, A. D-fap: Dual-factor authentication protocol for mobile cloud connected devices. J. Sens. Actuator Netw. 2020, 9, 1. [CrossRef]
- 146. Baig, M.M.; GholamHosseini, H.; Connolly, M.J. Mobile healthcare applications: System design review, critical issues and challenges. *Australas. Phys. Eng. Sci. Med.* 2015, *38*, 23–38. [CrossRef]
- 147. Chen, H.; Chiang, R.H.; Storey, V.C. Business intelligence and analytics: From big data to big impact. *Mis Q.* **2012**, 1165–1188. [CrossRef]
- 148. Jiang, F.; Jiang, Y.; Zhi, H.; Dong, Y.; Li, H.; Ma, S.; Wang, Y.; Dong, Q.; Shen, H.; Wang, Y. Artificial intelligence in healthcare: Past, present and future. *Stroke Vasc. Neurol.* 2017, *2*, 230–243. [CrossRef]

- 149. Magrabi, F.; Ammenwerth, E.; McNair, J.B.; de Keizer, N.F.; Hypponen, H.; Anen, P.N.; Rigby, M.; Scott, P.J.; Vehko, T.; Wong, Z.S.-Y.; et al. Artificial intelligence in clinical decision support: Challenges for evaluating ai and practical implications. *Yearb. Med. Inform.* 2019, 28, 128–134. [CrossRef]
- 150. ElSaadany, M. A Novel Iot-Based Wireless System to Monitor Heart Rate. Ph.D. Thesis, Miami University, Oxford, OH, USA, 2017.
- Lin, F.; Song, C.; Zhuang, Y.; Xu, W.; Li, C.; Ren, K. Cardiac scan: A non-contact and continuous heart-based user authentication system. In Proceedings of the 23rd Annual International Conference on Mobile Computing and Networking, Salt Lake City, UT, USA, 16–20 October 2017; pp. 315–328.
- 152. Casas, J.; Irfan, B.; Senft, E.; Gutierrez, L.; Rincon-Roncancio, M.; Munera, M.; Belpaeme, T.; Cifuentes, C.A. Social assistive robot for cardiac rehabilitation: A pilot study with patients with angioplasty. In Proceedings of the Companion of the 2018 ACM/IEEE International Conference on Human-Robot Interaction, Chicago, IL, USA, 5–8 March 2018; pp. 79–80.
- 153. Geurts, E.; Hansen, D.; Dendale, P.; Coninx, K. Impact of a mobile cycling application on cardiac patients' cycling behavior and enjoyment. In Proceedings of the 11th PErvasive Technologies Related to Assistive Environments Conference, Corfu, Greece, 26–29 June 2018; pp. 257–264.
- 154. El-kenawy, E.-S.M.; Eid, M.; Ibrahim, A. An advanced patient health monitoring system. J. Comput. Sci. Inf. Syst. 2021, 17, 1–7.
- Savaridass, M.P.; Ikram, N.; Deepika, R.; Aarnika, R. Development of smart health monitoring system using internet of things. *Mater. Today Proc.* 2021, 45, 986–989. [CrossRef]